

# FROM LABORATORY TO ROAD

A COMPARISON OF OFFICIAL AND REAL-WORLD FUEL CONSUMPTION AND  $CO_2$  VALUES FOR PASSENGER CARS IN EUROPE, THE UNITED STATES, CHINA, AND JAPAN

Uwe Tietge, Sonsoles Díaz, Zifei Yang, Peter Mock



www.theicct.org communications@theicct.org

# ACKNOWLEDGMENTS

The authors thank the internal and external reviewers of this report for their guidance and constructive comments, with special thanks to Maya Ben Dror, Liping Kang, and Lanzhi Qin (Innovation Center for Energy and Transportation); Jeff Alson and Roberts French (EPA); Iddo Riemersma; John German, Hui He, and Dan Rutherford (ICCT). The authors are grateful to the following individuals and organizations for contributing data and background information: Rick Goeltz, Janet Hopson (Oak Ridge National Laboratory); Xiong Zhang, Hongbo Sun (XiaoXiongYouHao); Masayoshi Ishihara (e-nenpi.com); and all who provided data for the 2016 update of the *From Laboratory to Road* study.

For additional information:

International Council on Clean Transportation Europe Neue Promenade 6, 10178 Berlin +49 (30) 847129-102

communications@theicct.org | www.theicct.org | @TheICCT

 $\ensuremath{\mathbb{C}}$  2017 International Council on Clean Transportation

Funding for this work was generously provided by the ClimateWorks Foundation and Stiftung Mercator.

# EXECUTIVE SUMMARY

The majority of all passenger cars sold in 2015 were subject to carbon dioxide  $(CO_2)$  or fuel economy standards. The six largest vehicle markets and nine of the top 11 have implemented such standards to curb fuel consumption and greenhouse gas emissions of light-duty vehicles. Leading markets have set passenger vehicle standards for 2020 and beyond, driving the decarbonization of road transportation and setting clear long-term targets for automakers. As a result, average  $CO_2$  emission values of new cars are declining on a global scale.

While fuel efficiency standards thus sound like a success, there is evidence of a growing divergence (or "gap") between official and real-world CO<sub>2</sub> values in a number of markets, implying that laboratory measurements are increasingly overestimating the fuel efficiency of cars. The ICCT analyzes the gap between official and real-world CO<sub>2</sub> emission values of European passenger cars in a series of studies termed *From Laboratory to Road*. This study extends the analysis beyond the borders of Europe and includes other major vehicle markets.

This study is based on real-world fuel consumption data for more than 1.5 million passenger cars in the European Union (EU), the United States, China, and Japan. Figure ES-1 indicates that the divergence between the official and real-world  $CO_2$  emission values has increased over time in all regions. The EU, which uses the New European Driving Cycle (NEDC) for vehicle testing, is the market with the steepest annual growth in the gap between 2001 and 2014, while Japan and the United States experienced smaller increases. Japan phased out the 10-15 mode cycle in the 2008-2011 time frame, replacing it with the JC08 cycle. The United States is the only market with two sets of  $CO_2$  values: the more limited two-cycle-based Corporate Average Fuel Economy (CAFE) values are used for regulatory purposes, while the U.S. Environmental Protection Agency (EPA) provides more inclusive five-cycle-based fuel economy label values to inform consumers about the real-world on-road performance of vehicles. The label values were found to offer the most realistic fuel consumption figures in the analysis. The label values are the basis for the adjustment factor used by U.S. regulators to convert CAFE values to real-world estimates during fuel economy standard rulemakings.



**Figure ES- 1.** Divergence between official and real-world  $CO_2$  emissions for new passenger cars in the EU, the United States, China, and Japan.

Unless regulators handle the growing gap appropriately, the divergence between official and real-world  $CO_2$  emission values will continue to dilute fuel efficiency policies. Figure ES-2 plots average official and estimated real-world  $CO_2$  emission values of passenger cars in the four regions under study. The EU and China appear to have made little progress in reducing on-road  $CO_2$  emission values after 2008, as the gap rapidly grew in both markets. The United States has the highest on-road  $CO_2$  emission values, but is reducing these values at a faster rate than the EU and China. Moreover, to compensate for the growing gap between CAFE and real-world  $CO_2$  values, U.S. regulators apply and periodically update adjustment factors in the impact assessments that accompany fuel economy rulemakings. Japan stands out with the lowest official and real-world  $CO_2$  emission values due to a light, efficient fleet and a comparatively low growth in the gap. Overall, the decoupling of official and real-world values in Figure ES-2 illustrates that the growing gap is a substantial obstacle to reducing  $CO_2$  emissions on the road, and must therefore be addressed when designing fuel efficiency policies.



**Figure ES-2.** Official and real-world CO<sub>2</sub> emission values for new passenger cars in the EU, the United States, China, and Japan.

In addition to studying the gap, this study summarizes vehicles testing procedures and policy frameworks in the four regions. A side-by-side comparison of procedures and divergence estimates highlights some aspects that are key to effective CO<sub>2</sub> or fuel economy standards:

- Independent retesting: Independent retesting of laboratory measurements was identified as a best practice. All markets have some form of compliance program in place, but the United States has the most extensive program, covering the full lifetime of vehicles by verifying coastdown measurements, testing production-line vehicles, and conducting in-use surveillance tests.
- » Policy enforcement: The comparatively low growth in the U.S. gap indicates that stringent policy enforcement, such as levying penalties on manufacturers that misstate fuel economy values, acts as a deterrent to gaming. In contrast, the EU has

seen the largest growth in the gap from 2001 to 2014 and lacks a central authority to issue vehicle recalls and to impose financial penalties.

- » **Real-world standards:**  $CO_2$  and fuel economy standards should be based on test values that, on average, correspond to real-world measurements. Policies that fail to account for the divergence will overestimate fuel savings and climate change mitigation benefits. Using an adjustment factor to approximate on-road values, as is done in the United States, is an approach that does not require extensive overhauls of vehicle testing procedures to account for real-world  $CO_2$  emissions. Another approach for measuring on-road emissions is using portable emissions measurement system (PEMS) equipment. On-road tests using PEMS are currently only being conducted for nitrogen oxides ( $NO_x$ ) and particulate number emissions as part of the real-driving emissions (RDE) procedure, but these kinds of measurements could be used to monitor on-road  $CO_2$  emissions and to test the on-road conformity of vehicles in use. More realistic test cycles and more rigorous testing procedures (e.g., including auxiliary equipment, using stock tires, and using standard engine and transmission calibrations) for laboratory testing could also furnish more realistic  $CO_2$  values.
- Real-world measurements: Measuring real-world fuel consumption is a key recommendation because these data are needed to evaluate the efficacy of CO<sub>2</sub> and fuel economy standards. Bulk on-road fuel consumption data can be measured using web services. As the only government-run example of such services, the MyMPG tool on FuelEconomy.gov stands out as a best-practice example. These data can be used to estimate fleet-wide real-world CO<sub>2</sub> emission values and gauge policy impacts. Using data loggers connected to vehicles' on-board diagnostics ports is another option for real-world fuel consumption data collection (see Posada & German, 2013).
- Consumer information: Consumers need access to realistic fuel consumption values to make informed decisions when buying vehicles. U.S. EPA window label values demonstrate that it is possible to produce fuel consumption values that, on average, are representative of real-world performance. The FuelEconomy.gov website stands out as a best-practice example of consumer information because it combines realworld measurements, realistic fuel consumption values, and information on efficient driving in one portal.

The growing divergence between official and on-road CO<sub>2</sub> emission values is troubling since it represents a decoupling of regulated metrics and real-world impacts. Good practices covered in this study illustrate that solutions are available to close or at least manage the gap. Differences in the development of the gap in Europe and the United States illustrate that effective policies and rigorous enforcement can change the trajectory of the gap: In the EU, the gap started growing at a faster rate after CO<sub>2</sub> standards were introduced in 2009, while the U.S. gap slowed down after fuel economy standards were reintroduced in 2012, largely due to a suite of policy measures including testing of production vehicles and in-use testing. This study aims to facilitate the transfer of such good practices and hard-won policy insights across markets.

# TABLE OF CONTENTS

Ex	Executive Summaryi						
Ak	brev	iations	3	vi			
1.	Intro	ductio	n	1			
2.	Vehi	cle tes	ting and regulatory frameworks	4			
	2.1	Light-	duty vehicle definitions	4			
	2.2	Passe	nger car characteristics	7			
	2.3	EU		8			
		2.3.1	Vehicle testing	8			
		2.3.2	CO <sub>2</sub> targets	8			
		2.3.3	Policy framework and enforcement	9			
	2.4	United	d States	10			
		2.4.1	Vehicle testing	10			
		2.4.2	Fuel economy and CO <sub>2</sub> emission targets	12			
		2.4.3	Policy framework and enforcement	13			
	2.5	China		15			
		2.5.1	Vehicle testing	15			
		2.5.2	Fuel economy and $\rm CO_2$ emission targets	15			
		2.5.3	Policy framework and enforcement				
	2.6	Japar	1	17			
		2.6.1	Vehicle testing	17			
		2.6.2	Fuel economy and $\rm CO_2$ emission targets				
		2.6.3	Policy framework and enforcement				
	2.7	Summ	nary of vehicle testing and CO <sub>2</sub> standards				
3.	Liter	ature	review	22			
	3.1	EU					
	3.2	United	d States	23			
	3.3	China		23			
	3.4	Japar	1	23			
4.	Data	analy	sis	25			
	4.1	Europ	e: Multiple sources	25			
	4.2	United	d States: FuelEconomy.gov				
	4.3	China	: XiaoXiongYouHao				
	4.4	Japar	n: e-nenpi.com				
	4.5	Data	comparison				
5.	Disc	ussion		41			
	5.1	Comp	parison of results and previous research	41			
	5.2	Reasc	ons for growing divergence estimates	41			
		5.2.1	Decreasing official CO <sub>2</sub> values	41			

5.2.2 Driver behavior						
5.2.3 Vehicle technologies						
5.2.4 Vehicle testing and policy frameworks						
5.2.5 Summary						
Policy implications						
Good practices for vehicle testing and policy frameworks						
Limitations and future research	47					
clusions and recommendations	50					
References						
Appendix I: Fuel economy charts						
dix I: Fuel economy charts	58					
n	<ul> <li>5.2.2 Driver behavior</li> <li>5.2.3 Vehicle technologies</li> <li>5.2.4 Vehicle testing and policy frameworks</li> <li>5.2.5 Summary</li> <li>Policy implications</li> <li>Good practices for vehicle testing and policy frameworks</li> <li>Limitations and future research</li> </ul>					

# ABBREVIATIONS

2WD	two-wheel drive
4WD	four-wheel drive
B7	diesel blend containing up to 7% biodiesel (volume/volume)
CAFC	Corporate Average Fuel Consumption
CAFE	Corporate Average Fuel Economy
CAGR	compound annual growth rate
CARB	California Air Resources Board
CO <sub>2</sub>	carbon dioxide
CoP	conformity of production
CVT	continuously variable transmission
E10	10% ethanol/90% gasoline blend (volume/volume)
E5	5% ethanol/95% gasoline blend (volume/volume)
EEA	European Environment Agency
EU	European Union
EUDC	Extra Urban Driving Cycle
FTP-75	Federal Test Procedure
g	gram
GVW	gross vehicle weight
HEV	hybrid electric vehicle
HWFET	Highway Fuel Economy Test
ICCT	International Council on Clean Transportation
IUCP	In-Use Confirmatory Program
IUVP	In-Use Verification Program
km	kilometer
I	liter
LCV	light commercial vehicle
LDV	light-duty vehicle
m	meter
M1	passenger car with a gross vehicle weight not exceeding 3.5 tons
M1G	off-road passenger cars
M2	light commercial passenger vehicles
METI	Japanese Ministry of Economy, Trade and Industry
MIIT	Chinese Ministry of Industry and Information Technology
MLIT	Japanese Ministry of Land, Infrastructure, Transport and Tourism
mpg	miles per gallon
MPV	multi-purpose vehicle
MY	model year
N1	light commercial vehicle with a gross vehicle weight not exceeding 3.5 tons

- NEDC New European Driving Cycle
   NHTSA National Highway Traffic Safety Administration
   NO<sub>x</sub> nitrogen oxides
   NTSEL National Traffic Safety and Environment Laboratory
   PEMS Portable emissions measurement system
- PHEV plug-in hybrid electric vehicle
- RDE Real-driving emissions
- SEA Selective Enforcement Audit
- SUV sport utility vehicle
- U.S. United States of America
- U.S. EPA United States Environmental Protection Agency
- UK United Kingdom
- UNECE United Nations Economic Commission for Europe
- WLTP Worldwide Harmonized Light Vehicles Test Procedure

# 1. INTRODUCTION

Most major vehicle markets in the world have carbon dioxide  $(CO_2)$  or fuel economy standards in place to curb fuel consumption and greenhouse gas emissions of light-duty vehicles (see Box 1.1 for definitions of metrics). Approximately 78% of all vehicles sold in 2015 were subject to  $CO_2$  or fuel consumption standards (ICCT, 2016a). Leading markets have set passenger vehicle standards for 2020 and beyond, driving the decarbonization of road transportation and setting clear long-term targets for automakers.

## **BOX 1.1: OVERVIEW OF VEHICLE EFFICIENCY METRICS**

Fuel consumption refers to fuel consumed per unit of distance traveled. Fuel economy refers to distance traveled per unit of fuel consumed. The two metrics are therefore inversely related: As fuel consumption decreases, fuel economy increases.  $CO_2$  emissions are directly proportional to fuel consumption and depend on the carbon intensity of the fuel. Fuel economy values are common in the United States, expressed as miles per gallon (mpg), and in Japan, expressed as kilometer per liter (km/l). Fuel consumption values, expressed as liters per 100 kilometers (l/100 km), are common in the EU and in China. All markets use  $CO_2$  emission figures for some purposes, which are expressed either as grams of  $CO_2$  per kilometer (g/km) or grams of  $CO_2$  per mile (g/mi).



As a result of CO<sub>2</sub> and fuel economy standards, greenhouse gas emissions from new light-duty vehicles (LDVs) are declining in major vehicle markets. The United States has a long history of fuel economy targets reaching back to 1975. The most recent fuel economy target requires that average CO<sub>2</sub> emission values of passenger cars fall below approximately 97 g CO<sub>2</sub>/km by 2025 (all values normalized to the New European Driving Cycle), which translates to a 4.8% annual reduction. The EU agreed on a 2015 fleet-average target of 130 g CO<sub>2</sub>/km in 2008. As a result, average CO<sub>2</sub> emission values of new European passenger cars have declined by roughly 4% per year, up from 1% before the standards were introduced. Major Asian vehicle markets such as China (93 g CO<sub>2</sub>/km by 2025) and Japan (122 g CO<sub>2</sub>/km by 2020) also have CO<sub>2</sub> standards or fuel economy targets in place (see Figure 1).



**Figure 1.** Global new passenger car CO<sub>2</sub> emissions and fuel consumption, normalized to NEDC, including future targets (source: ICCT, 2016b; methodology: Kühlwein, German, & Bandivadekar, 2014).

Although fuel efficiency standards have been successful at reducing declared  $CO_2$  and fuel consumption values, these values are measured in laboratories under controlled circumstances and rarely reflect measurements of average emissions from on-road driving. However, to achieve real-world benefits, official  $CO_2$  values measured in the laboratory must translate into on-road emission reductions and fuel savings.

The International Council on Clean Transportation (ICCT) began to investigate the divergence (or "gap") between European official and real-world  $CO_2$  and fuel consumption values in 2012 (Mock, German, Bandivadekar, & Riemersma, 2012). Results from these investigations are summarized in a series of studies termed *From Laboratory to Road*. These studies have found evidence of a growing divergence between official and real-world  $CO_2$  values for passenger cars in Europe. Based on data for approximately 1 million vehicles from 13 data sources and seven countries, the 2016 study found that the divergence increased from approximately 9% in 2001 to 42% in 2015 (Tietge et al., 2016). The growing divergence counteracts the EU  $CO_2$  standards and dilutes their intended benefits, which include mitigating climate change, reducing oil imports, and reducing consumers' fuel expenses.

This study reaches beyond Europe's borders and takes a global perspective on realworld  $CO_2$  emissions from passenger cars by studying three additional major vehicle markets: the United States, China, and Japan. Roughly two-thirds of new passenger cars in the world in 2015 were registered in one of these four markets. Standards set by the EU and the United States also affect fuel economy and  $CO_2$  standards in other markets, since European and U.S. regulations and test procedures are used elsewhere. For instance, Brazil, Canada, Mexico, and South Korea use U.S. test cycles, while India and China employ test procedures similar to EU regulations. Findings from this study therefore also have implications for other markets. The objective of this study is to compare the divergence between on-road and official CO<sub>2</sub> values in the four studied markets in order to evaluate the efficacy of different regulations and to distill good practices for policy design. Some studies of regional on-road environmental performance of cars have been conducted before (Ding, Ben Dror, Kang, & An, 2015; Greene et al., 2015; Huo, Yao, He, & Yu, 2011; Ntziachristos et al., 2014; Qin, Ben Dror, Kang, Sun, & An, 2016; Tietge et al., 2015), but studies of global scope have traditionally relied on real-world fuel consumption estimates based on fuel sales (e.g., Schipper, 2008; Schipper & Tax, 1994). This study provides the first global comparison of the divergence between on-road and official CO<sub>2</sub> emission values.

To provide estimates of the divergence between official values and real-world performance, this study compares large samples of on-road fuel consumption measurements. Findings are presented in terms of divergence between official and real-world  $CO_2$  emission values. The *divergence* is defined as the difference between real-world and official values, expressed as a percentage of the official value. In other words, a 30% gap implies that real-world  $CO_2$  emissions are 30% higher than measured in the laboratory. It should be noted that calculating the divergence based on  $CO_2$  emission values yields different results than using fuel economy figures, as is common practice in Japan and the United States (see Section 4.5 for a discussion of the relationship between fuel economy and fuel consumption divergence). Figures using fuel economy are presented in Appendix I to make the findings more accessible to readers from these markets. While findings are presented in terms of  $CO_2$  emission values, the same divergence estimates also apply to fuel consumption as virtually all of the carbon in the fuel is converted to  $CO_2$  during combustion. The terms *fuel consumption* and  $CO_2$  *emissions* are used interchangeably throughout the study.

The remainder of this study is organized in five parts. Section 2 provides an overview of the four vehicle markets and vehicle testing frameworks. Section 3 summarizes findings from previous studies on the real-world gap in the four markets. Section 4 presents real-world fuel consumption data from the four markets, quantifies the divergence between official and real-world values, and compares the findings. Section 5 discusses the relationship between the design of vehicle testing frameworks and the divergence between testing and on-road emissions, and distills good practices identified in the analysis. Lastly, Section 6 summarizes the findings and presents policy recommendations.

# 2. VEHICLE TESTING AND REGULATORY FRAMEWORKS

#### **BOX 2.1: LABORATORY VEHICLE TESTING**

To measure fuel consumption and  $CO_2$  emissions in laboratories, vehicles are placed on chassis dynamometers. Chassis dynamometers can be compared to large treadmills, allowing the vehicle to remain stationary while the wheels spin. The resistance placed on the rollers of the chassis dynamometer by an electronic controller simulates the inertia effects due to the weight of the vehicle and the aerodynamic drag and rolling resistance forces acting on a vehicle during on-road operation. These two effects are commonly referred to as the *road load*. Road load is measured during coastdown testing, where vehicles are accelerated to a certain speed and then coast in neutral. The time it takes for the vehicle to decelerate is used to estimate the road load force acting upon the vehicle.

This section explores the different vehicle markets, testing procedures, and regulatory frameworks in the regions studied in the analysis.

#### 2.1 LIGHT-DUTY VEHICLE DEFINITIONS

The regions covered in this study use varying definitions of LDVs. Definitions vary in terms of vehicle size, vehicle weight, intended use, and body style.

In the EU, LDVs include vehicles classified as passenger cars, referred to as *M1 vehicles*, and light commercial vehicles, referred to as *N1 vehicles*. Passenger cars are intended to carry passengers, do not exceed 3,500 kg gross vehicle weight (GVW), and have no more than eight passenger seats (European Parliament, 2007). Light commercial vehicles are intended to carry goods and do not exceed 3,500 kg GVW. Determining whether a vehicle is intended for passenger or goods transportation is subject to national legislation, so that a certain vehicle model may be registered as M1 in one member state and as N1 in another. See Figure 2 for an overview and market shares of the different vehicle categories in the EU.



**Figure 2.** Light-duty vehicle shares of 2014 registrations by vehicle category and segment in the EU (source: Mock, 2015).

In the United States, vehicles are primarily classified not according to their intended use, but rather according to their GVW and body type. For the purposes of fuel economy certification, the U.S. Environmental Protection Agency (EPA) classifies vehicles as light-duty (GVW  $\leq$  3,900 kg) and heavy-duty (GVW > 3,900 kg), although SUVs and passenger vans with GVW ratings between 3,900 and 4,500 kg (medium-duty passenger vehicles) have recently been added to the light-duty fuel economy regulation (U.S. EPA, 2010a). The U.S. LDV category comprises passenger cars and light-duty trucks. The regulatory definitions of cars and trucks have evolved over time and are currently based on the ability of the vehicle to perform certain functions (e.g., provide temporary living quarters, transport property on an open bed) (U.S. EPA, 2015b). The term *passenger car* refers to vehicles with car-like body types and a GVW up to 3,900 kg, as well as small and mid-size two-wheel (2WD) drive SUVs with a GVW up to 2,700 kg. Light trucks are all vans and four-wheel drive (4WD) SUVs up to 4,500 kg GVW, plus cargo vans and pick-up trucks up to 3,900 kg GVW. See Figure 3 for an overview and market shares of the different vehicle categories in the United States.



**Figure 3.** Light-duty vehicle shares of model year 2014 by vehicle category and segment in the United States (source: U.S. EPA, 2015b).

In China, according to the standard GB 15089-2001 introduced in 2001, the LDV category comprises passenger cars, classified as M1, and light commercial vehicles, classified as N1 or M2 (He & Tu, 2012). Similar to the EU M1 definition, China's passenger cars are vehicles whose main function is to carry people, have nine or fewer seats, and weigh no more than 3,500 kg. Fuel consumption standards further divide M1 vehicles into M1 and M1G, where the M1G category refers to off-road-capable passenger cars and M1 comprises the remaining passenger car body types. M1G vehicles were held to a more lenient set of standards than M1 vehicles until 2011. The term *light commercial vehicle* refers to both light commercial passenger vehicles (M2 category) and light commercial cargo vehicles (N1 category) (Tu, Zou, & He, 2014). The M2 category refers to passenger vehicles with more than nine seats and a GVW less than 5,000 kg. However, only lighter M2 vehicles with a GVW below 3,500 kg are considered light commercial vehicles. N1 vehicles are primarily designed for the transportation of goods and, as all LDVs, have a GVW of 3,500 kg or less. See Figure 4 for an overview and market shares of the different vehicle categories in China.



**Figure 4.** Light-duty vehicle shares of 2014 registrations by vehicle category and segment in China (data provided by Segment Y).

Lastly, Japanese light-duty fuel efficiency standards classify LDVs into the following three categories: passenger cars, small buses, and freight vehicles. *Passenger cars* are defined as vehicles designed to carry 10 passengers or fewer, whereas *small buses* are described as passenger vehicles with a capacity of 11 passengers or more and a GVW below 3,500 kg. Freight vehicles are further divided into three categories: mini, light-weight, and medium-weight. *Mini freight vehicles* are defined as cargo vehicles whose length, width, and height do not exceed 3.4 m, 1.48 m, and 2.0 m respectively, and whose engine capacity does not exceed 0.66 l. Light and medium-weight freight vehicles have a maximum a GVW of 1,700 kg and 3,500 kg respectively. See Figure 5 for an overview and market shares of the different vehicle categories in Japan.



**Figure 5.** Light-duty vehicle shares of 2014 registrations by vehicle category and segment in Japan. Small buses account for about 0.1% of the market and are therefore not displayed in the figure (source: JAMA, 2015).

While all four regions generally define LDVs by weight (GVW of less than approximately 3,500 kg), light-duty categories or segments are not always comparable across regions. For instance, a 4WD SUV intended for passenger transportation is subject to passenger car regulations in the EU and China, but is subject to light truck standards in the United States.

To produce comparable results, **the scope of this study is limited to passenger cars**. Passenger cars are defined here as follows: M1 car-like segments in the EU; cars plus 2WD SUVs with GVW below 2,700 kg in the United States; M1 car-like segments and M1G vehicles in China; and passenger cars in Japan. Van-like models were excluded from both the EU and China analyses, because it was not possible to determine whether such vehicles were registered as M1 or N1 vehicles. The Japanese on-road fuel consumption data did not contain any information on vehicle category or segment, so it was not possible to ensure that vehicles other than car-like passenger cars were excluded from the analysis. In short, to ensure comparability of results, the analysis focuses on vehicles that are registered as passenger cars and have body types resembling stereotypical cars and SUVs.

## 2.2 PASSENGER CAR CHARACTERISTICS

Even though the study only focuses on passenger cars, the markets for these cars vary substantially in terms of basic vehicle characteristics. Table 1 provides an overview of average new car parameters in each of the four regions under study. While EU and China fleet specifications refer to new 2014 registrations, the U.S. data refers to model year 2014, indicating when a vehicle was marketed, and the values for Japan to sales year 2011 (more recent data was not available). Vehicle mass is provided as curb weight, which is commonly defined as the weight of the empty vehicle with standard equipment, all necessary operating consumables, and a fuel tank roughly filled to capacity. In the EU, however, vehicle mass is typically given as mass in running order, which includes the mass of the driver in addition to the curb weight. To account for the difference in the definition, a mass of 75 kg has been subtracted from the EU mass in running order value.

The EU and China are similar in terms of vehicle mass and engine power. U.S. passenger cars are the heaviest and most powerful on average, while Japanese cars are on the other end of the spectrum. The United States has the most powerful cars even when accounting for vehicle mass, with a power-to-mass ratio around 40% higher than the rest. Diesel vehicles are only popular in the EU, where they accounted for 53% of new vehicle registrations in 2014. The EU is also an exception regarding transmission types, as it is the only region where vehicles with automatic transmission do not account for the majority of car registrations. In terms of  $CO_2$  emissions, China and the United States stand out with the highest average values, while the EU and Japan average values are around 26% lower.

Region	EU	U.S.	China	Japan
Dating convention	Sales year	Model year	Sales year	Fleet year
Year	2014	2014	2014	2011
Curb weight (kg)	1,317	1,478	1,376	1,193
Power (kW)	90	148	101	78
Specific power (kW/kg)	0.070	0.100	0.073	0.065
Diesel (%)	53%	1%	2%	1%
Automatic transmission (%)	23%	96%	56%	99%
$CO_2$ (g $CO_2$ /km)	123	167	171	126

**Table 1.** Comparison of new passenger car characteristics.  $CO_2$  emission values normalized to NEDC (sources: Mock, 2015; U.S. EPA, 2015b).

## 2.3 EU

#### 2.3.1 Vehicle testing

In the EU, LDV CO<sub>2</sub> emission and fuel consumption figures are determined in laboratories using the United Nations Economic Commission for Europe (UNECE) R101 procedure and the New European Driving Cycle (NEDC), a cycle developed in the 1970s and last updated in 1998 (Kühlwein, 2016). The NEDC consists of four consecutive urban driving cycles, termed ECE-15, followed by an Extra Urban Driving Cycle (EUDC). While the ECE-15 represents urban driving with low vehicle speeds and engine load, the EUDC covers higher speeds, up to 120 km/h (see Figure 6). Before the testing begins, the vehicle is conditioned to an ambient temperature of 20°C to 30°C for at least six hours. During NEDC testing, auxiliary electric devices such as air conditioning or entertainment systems are turned off. Testing conditions (e.g., the high ambient temperature) and a number of tolerances and flexibilities—allowable tolerances for laboratory instruments, testing of so-called "golden vehicles", special test driving techniques, etc.—provide opportunities to produce particularly low CO<sub>2</sub> emission values during type-approval testing (Stewart, Hope-Morley, Mock, & Tietge, 2015).

Before the chassis dynamometer testing begins, resistance values are determined during coastdown testing. The coastdown procedure used to estimate road loads for NEDC testing also includes technical tolerances and imprecise definitions, indicative of the poor technical standards and imprecise instrumentation of the 1970s, but out of date compared with modern technologies and standards (Kühlwein, 2016). Manufacturers can exploit a number of imprecisions, allowing for pretreatment of tires by baking or shaving them, optimizing aerodynamics, opening brake calipers, or carefully selecting test tracks with smooth and hard road surfaces, among others (Kadijk et al., 2012; Kühlwein, 2016). The problems with the out-of-date requirements are compounded by the lack of any oversight of the manufacturer testing.



Figure 6. Driving schedule of the NEDC

#### 2.3.2 CO, targets

The first binding CO<sub>2</sub> target in the EU was set in 2009 and required average CO<sub>2</sub> emission values of new passenger cars to fall below 130 g/km by 2015 (Regulation (EC) No 443/2009, 2009). A second target of 95 g CO<sub>2</sub>/km for 2020, with a 1-year phase-in period, was set in 2014 (European Commission, 2014). The European Commission (2016a) is also working on post-2020 CO<sub>2</sub> standards, including an intermediate target before 2030.

All vehicle manufacturers must comply with CO<sub>2</sub> standards, although targets are adjusted by vehicle mass such that manufacturers of heavier cars are allowed higher CO<sub>2</sub> emissions, whereas manufacturers of lighter cars must meet more stringent targets. The pre-production type-approval test results are weighted by vehicle sales to calculate

a manufacturer's fleet average  $CO_2$  emission values. If a manufacturer fails to comply with its  $CO_2$  target, it must pay an excess emissions premium, which can easily amount to multimillion-euro penalties. Up until 2019, the premiums, per vehicle, are: 5 euros for the first g/km of exceedance; 15 euros for the second g/km; 25 euros for the third g/km; and 95 euros for each subsequent g/km (European Commission, 2016c). A number of provisions are in place to help manufacturers meet their targets, and include multipliers ("super-credits") for low-carbon vehicles, emission credits for innovative off-cycle efficiency technologies ("eco innovations"), rules allowing manufacturers to enter pools to jointly meet targets, and separate targets or derogations for smaller manufacturers, among others (for more details, see Mock, 2014).

According to official values, so-called type-approval values<sup>1</sup>,  $CO_2$  emissions from new cars in the EU have rapidly decreased from 168 g  $CO_2$ /km in 2001 to 120 g  $CO_2$ /km in 2015. The rate of reductions increased noticeably after  $CO_2$  standards were introduced in 2009 (see Figure 7), illustrating that the 2015 target was effective at driving down  $CO_2$  emission and fuel consumption values, at least on paper.



Figure 7. Development of average type-approval CO, emission values for new passenger cars in the EU.

#### 2.3.3 Policy framework and enforcement

Type-approval  $CO_2$  emissions testing in the EU is done primarily with laboratory tests of pre-production vehicles based on the NEDC. After laboratory testing is completed, conformity of production (CoP) certification requires manufacturers to demonstrate that each vehicle is manufactured to the approved specifications, which typically can be proven by using quality-management systems (Mock & German, 2015). Manufacturers must also retest randomly chosen vehicles from the assembly line, and  $CO_2$  emissions may not deviate from the type-approval value by more than 8%. On-road tests using portable emissions measurement systems (PEMS) are currently only being conducted for nitrogen oxides ( $NO_x$ ) and particulate number emissions as part of the real-driving emissions (RDE) procedure (Franco & Mock, 2015). The European Commission (2016a) is investigating the feasibility of measuring real-world  $CO_2$  emissions. The current

<sup>1</sup> *Type-approval* refers to the process of vehicle certification, that is, demonstrating the roadworthiness of a vehicle type, in China and in the European Union.

laboratory testing procedure based on the NEDC will be replaced by the Worldwide Harmonized Light Vehicles Test Procedure (WLTP) beginning in 2017. The WLTP will introduce a more dynamic test cycle and more realistic test parameters related to vehicle weight and ambient temperature, among others. It will also introduce improvements to coastdown testing. Nevertheless, the WLTP is not expected to align official and realworld measurements by itself (Stewart et al., 2015).

The current European type-approval framework has systemic flaws related to how different organizations in the type-approval chain interact. First, car manufacturers pay technical services to conduct vehicle tests. Technical services therefore have an incentive to produce favorable test results to attract business from car manufacturers (ICCT, 2016c). Competition between different technical services may also create a "race to the bottom." Second, important test parameters, most notably road-load coefficients, are not independently verified or even publicly available (Kühlwein, 2016). Lastly, European regulators and member states do not have the authority to impose financial penalties for noncompliance, and only the member state that issued the type approval for a particular model variant can revoke the certificate. This market structure and the lack of transparency provide opportunities for car manufacturers to furnish unrealistic CO<sub>2</sub> emission values.

The impact assessment accompanying the 2020  $CO_2$  standards acknowledges this issue. Some cost scenarios in the impact assessment, including the central scenario, assume that the cost of complying with the 2020 standard would be lower than industry estimates because the decline in average  $CO_2$  emission values from 2002 to 2009 was achieved without significant deployment of new technologies, which are the major driver of compliance costs. The impact assessment also uses a factor of 1.195 (corresponding to a 19.5% gap) to convert type-approval values to real-world estimates. This factor was based on a 2006 study (see Smokers et al., 2006) and underestimates the gap between real-world and official CO2 values, which has grown to a level of 42% for new vehicles in 2015 (see Section 4.1).

A proposal by the European Commission (2016b) aims to overhaul the legal framework of the European type-approval scheme to address some of the aforementioned shortcomings. The proposal would (1) require EU member states to perform market surveillance testing; (2) allow EU member states to take measures, such as recalls and fines, against noncompliant vehicles; (3) create an information exchange for enforcement expertise; and (4) break financial ties between vehicle manufacturers and organizations conducting the type-approval testing and provide funding for market surveillance activities (Franco, 2016; Yang & Muncrief, 2017).

## 2.4 UNITED STATES

#### 2.4.1 Vehicle testing

Two types of fuel economy values are available for light-duty vehicles in the United States: Corporate Average Fuel Economy (CAFE) values are used for fuel economy standards, while Fuel Economy Label values (here referred to as *U.S. EPA label values*) are presented to consumers at the point of purchase.

CAFE values are based on laboratory measurements of exhaust emissions during two driving cycles, the FTP-75 and the U.S. EPA Highway Fuel Economy Test Cycle (HWFET):

» FTP-75: The FTP-75 is used to estimate urban fuel consumption of vehicles. The cycle consists of three phases (see Figure 9): the cold start phase, the stabilized phase, and the hot-start phase. The hot-start phase is conducted after a 10-minute

break following the stabilized phase. The average speed during the test is 34 km/h and ranges up to 91 km/h.

» HWFET: The HWFET was developed to measure fuel economy during highway driving. This is a hot-start test. While the maximum speed (97 km/h) is similar to the FTP-75, average speeds are considerably higher at 78 km/h.

The two test cycles are respectively weighted 55% and 45% in the calculation of the fuel economy value.

U.S. EPA label values are meant to be representative of fuel consumption during realworld driving. The values are presented at the point of purchase so that consumers can compare the fuel economy and fuel costs of different vehicles. Until model year 2007, the label value was based on adjusting the city (FTP) fuel economy value downward by 10% (equivalent to an 11% increase in fuel consumption) and the highway (HWFET) value downward by 22% (equivalent to a 28% increase in fuel consumption) (U.S. EPA, 2006). Since model year 2008, a new procedure, termed the *five-cycle method*, includes additional laboratory fuel consumption tests based on three cycles (see Figure 9):

- » US06: The US06 Supplemental Federal Test Procedure was developed to represent more aggressive driving, including higher speeds and higher acceleration rates than the FTP-75 test cycle.
- » SC03: The SC03 Supplemental Federal Test Procedure was developed to represent engine load resulting from the use of air conditioning, with vehicles tested at high ambient temperatures (35°C), full sun load, and air conditioning turned on.
- » Cold FTP: The cold FTP refers to a FTP-75 test at reduced ambient temperatures, specifically -6.7°C, instead of 20°C-30°C.

The weighting of the five cycles in the calculation of average urban, highway, and combined fuel economy values takes into consideration average vehicle usage patterns and ambient conditions (U.S. EPA, 2006). On top of using the three more demanding cycles in the calculation of U.S. EPA fuel economy labels, the final fuel economy values are adjusted downward by 9.5% (corresponding to a 10.5% increase in fuel consumption) to account for real-world factors that are not measured during laboratory testing, including fuel quality, tire pressure, wind, road gradient, etc. The U.S. EPA (2006) estimated that the five-cycle method reduced urban fuel economy values by 8% to 15% (corresponding to a 9% to 18% increase in fuel consumption) and reduced highway fuel economy values by 5% to 15% (corresponding to a 5% to 18% increase in fuel consumption) for most conventional vehicles compared with the procedure used before 2008.

In practice, the five-cycle values are commonly calculated from the FTP-75 and HWFET fuel economy values (see U.S. EPA, 2015c). This simplified approach can be used for all versions of a vehicle model as long as manufacturers can demonstrate that the derived five-cycle method yields comparable results to those achieved by the full five-cycle method for the certification vehicle (U.S. EPA, 2015b).



**Figure 8.** U.S. driving schedules used for the CAFE standards (FTP-75 & HWFET) and EPA label values (all cycles) (source: DieselNet, 2016).

#### 2.4.2 Fuel economy and CO, emission targets

The United States set its first fuel economy standard for LDVs in 1975, aiming to roughly double the average fuel economy of cars from 13.6 miles per gallon (mpg) (17.3 I/100 km) in 1974 to 27.5 mpg (8.6 I/100 km) by 1985 using the unadjusted CAFE test results, a target that was met with a one-year delay (DieselNet, 2013; NHTSA, 2014). A second set of standards was set for light trucks only for model years (MY) 2005 to 2011. A third stage required cars to increase fuel economy to 37.8 mpg (6.2 I/100 km or 140 g  $CO_2$ /km) and light trucks to meet a target of 28.8 mpg (8.2 I/100 km or 185 g  $CO_2$ /km) (ICCT, 2016a) by 2016. This third stage, covering MY 2012 to MY 2016, also adjusted targets for vehicle size, allowing manufacturers with larger vehicle

footprints to have lower fuel economy targets. Other flexibilities include credits for improvement in air conditioning systems, electric vehicles, and technologies that reduce  $CO_2$  emissions outside of vehicle testing, among others (DieselNet, 2013; U.S. EPA, 2010b). A fourth stage of standards set targets up to 2025: an estimated 56.2 mpg (4.2 I/100 km or 89 g  $CO_2$ /km) for cars and 37.8 mpg (6.2 I/100 km or 126 g  $CO_2$ /km) for light trucks (ICCT, 2016a).





#### 2.4.3 Policy framework and enforcement

A wide variety of policy enforcement measures are in place in the United States to ensure compliance with emission and fuel economy regulations. Figure 10 follows a vehicle through the U.S. EPA compliance program. The unit of analysis for most of this program is a test group, a group of vehicles or engines with similar technical and emission characteristics (for more details, see Mock & German, 2015).

During the preproduction phase to establish fuel economy label values, the U.S. EPA does confirmatory testing of about 15% of the vehicles tested by the manufacturers, who also submit their results to the U.S. EPA. If there is an ongoing offset between the two test results, U.S. EPA will increase the confirmatory testing rate until the manufacturer fixes the problem. The U.S. EPA also has extensive guidance on all aspects of vehicle testing, including defining how to determine the vehicle weight, accessories installed on the vehicle, selection of representative tires (and associated road-load), mileage accumulation adjustments, and how to test vehicles with driver-selectable devices.

During the vehicle design-and-build phase, the U.S. EPA may test vehicles off the assembly line to ensure that they are built in accordance with specifications of prototypes used for certification, so-called Selective Enforcement Audits (SEAs). SEAs have become less common over time: Failed audits were becoming rare by the mid-1980s, as they had triggered manufacturers to conduct extensive tests themselves (Mock & German, 2015). Road-load coefficients, which are also determined during

the vehicle design and build phase, are publicly accessible online, and the U.S. EPA conducts confirmatory coastdown tests on vehicles that are in use (Kühlwein, 2016).

For CAFE and CO<sub>2</sub> standard compliance, the U.S. EPA requires testing of vehicle configurations representing at least 90% of each manufacturer's actual production. This means that the manufacturers conduct another round of testing after the model year ends and they have final production numbers, also subject to U.S. EPA confirmatory testing. This eliminates problems with unrepresentative pre-production test vehicles and ensures that the test vehicles properly represent actual production. Note that this means that the final fuel economy values for each model type determined for standard compliance might be significantly different from the fuel economy label values that were based upon prototype vehicles and early calibrations.

Once vehicles enter the market, the In-Use Verification Program (IUVP) requires manufacturers to conduct chassis dynamometer tests on at least one in-use vehicle for each test group at low mileage (10,000 miles, or 16,000 km) and high mileage (50,000 miles, or 80,000 km). Emission exceedances during the IUVP can trigger an In-Use Confirmatory Program (IUCP) test, which requires manufacturers conduct further investigations. Failure of IUCP tests can lead to recalls. To ensure the accuracy of the manufacturers' testing, the U.S. EPA also randomly selects or targets suspicious test groups during in-use surveillance tests, which involve testing three to five vehicles from a test group to ensure compliance with declared values and emission standards.

In addition to monitoring vehicle testing, U.S. authorities account for the gap between regulated CAFE values and on-road measurements: the U.S. EPA uses a real-world adjustment factor in the impact assessment for fuel economy standards. For instance, for the third stage of standards (MY 2012 to MY 2016), CAFE fuel economy targets were adjusted downward by 20% (equivalent to an increase in fuel consumption by 25%), reflecting the average gap at the time (U.S. EPA, 2010c). The midterm evaluation of the fourth stage of standards for 2017 to 2025 increased the adjustment factor to 23% (equivalent to an increase in fuel consumption by 30%), in anticipation of a growing gap due to increased ethanol content in fuels (U.S. EPA, CARB, & NHTSA, 2016). The U.S. fuel economy regulations are thus accompanied by impact assessments that take into consideration the gap between CAFE and real-world values.<sup>2</sup>

Taken together, the U.S. EPA compliance program follows vehicles throughout their useful life and has mechanisms in place to ensure independent testing of vehicles and testing of representative vehicles (see Figure 10). In contrast to the EU, the regulator (the U.S. EPA) also has the power to issue vehicle recalls and to impose fines for non-compliance.

<sup>2</sup> It should be noted that the U.S. EPA adjustment procedure for the CO<sub>2</sub> and fuel consumption impact assessments could be improved. The U.S. EPA has found that the "gap" between test and real-world fuel economy is, in part, a function of the fuel economy of the vehicle, such that higher fuel economy vehicles have a higher gap. This is reflected in the fuel economy label adjustment formula, but instead of using this formula for the impact assessment, the U.S. EPA used the same adjustment factor (23% for the latest rule) for both the baseline fleet and the future fleet. When incorporating the average fuel economy, the gap for the average fleet mpg of about 28 in 2008-2010 yields a correction factor of about 23.5%, and the 2025 average fleet mpg of about 49 yields a correction factor of about 26.5%. The impact of this correction is much smaller than incorporating a factor for the gap into the impact assessments, but would still be an incremental improvement.



Figure 10. U.S. EPA compliance program for light-duty vehicles (Mock & German, 2015; U.S. EPA, 2008).

# 2.5 CHINA

## 2.5.1 Vehicle testing

China has followed the UNECE R101 to determine LDV fuel consumption figures in laboratories since the adoption of fuel consumption standards in 2004. The testing methodology is thus exactly the same as EU testing methods, using the four consecutive ECE-15 (urban) and a EUDC (extra-urban) of the NEDC with the same ambient temperature (see Figure 6).

To estimate road loads for chassis dynamometer testing, manufacturers can refer to predetermined values, so-called *cookbook values*, or conduct coastdown tests following the UNECE R83 procedure (the same as in the EU).

## 2.5.2 Fuel economy and CO<sub>2</sub> emission targets

China's first fuel consumption standards for passenger vehicles took effect in 2005 and applied only to domestic vehicles. The standards were implemented in two phases: Phase I took effect on July 1, 2005, and Phase II took effect on January 1, 2008, for new models and a year later for continued models. These first two phases set fuel consumption limits for individual vehicle models by weight category.

China then released the Phase III fuel consumption standards to regulate vehicles produced from 2012 on that applied to both domestic and imported vehicles. In contrast to the first two phases, the Phase III standards define corporate-average fuel consumption (CAFC) targets for each manufacturer in addition to per-vehicle targets. Similar to the EU  $CO_2$  regulations, manufacturers were required to meet an average fuel consumption target, which was adjusted by the vehicle curb weight distribution across the manufacturers' fleet, limiting overall fleet-average fuel consumption values to 6.9 I/100km by 2015.

The Phase IV standards, which were released in 2014, follow the same regulatory structure as Phase III, targeting a fleet-average fuel consumption of 5 I/100km in 2020. Phase IV includes a number of flexibility mechanisms, such as multipliers for new energy vehicles and ultra-low fuel consumption vehicles, off-cycle technology credits, and a phase-in period (He & Yang, 2014b). China now is working on future standards with a proposed target of 4 I/100km by 2025.

Figure 11 shows the development of the fuel consumption standards in China over time. The term *special vehicles* refers to vehicles with automatic transmission and three rows of seats or more for Phase I, II, and III standards, and refers to three rows of seats or more for the Phase IV standards.



**Figure 11.** Development of fuel consumption standards and fleet trends in China for regular vehicles (Reg.) and special vehicles (Spec.).

#### 2.5.3 Policy framework and enforcement

The enforcement of vehicle fuel consumption standards differs from the enforcement of conventional pollutant standards. The former is currently regulated by the Ministry of Industry, Information, and Technology (MIIT), whereas the Ministry of Environmental Protection regulates the latter.

The enforcement of fuel consumption standards in China mainly focuses on tests of pre-production vehicles in the laboratory. Manufacturers must test vehicles at a certified laboratory to conduct type-approval testing for LDVs and then submit the data with their application for type approval.

The regulation requires manufacturers to ensure CoP of produced vehicles, but manufacturers are not required to prove that vehicles meet certified fuel consumption levels. The regulatory agency has the authority to select vehicles from the production line to carry out a CoP test, but there is no evidence that proves such tests are carried out on a regular basis, nor that any noncompliance has been discovered by such tests. There is no in-use surveillance testing of vehicles to verify that vehicles meet the certified values throughout their useful life.

China enforces its fuel consumption standards with administrative, rather than financial, penalties. From November 1, 2014, on, if a manufacturer does not meet its CAFC target with only its conventional fuel passenger cars and also cannot meet a uniform 6.9 I/100km corporate fuel consumption target after counting in its New Energy

Vehicles, which include battery electric vehicles and plug-in hybrid electric vehicles (PHEVs), MIIT will "name and shame" the manufacturer. In addition, MIIT can suspend the new type approval (vehicle product catalog) application of noncompliant manufacturers, ban or limit the production of new models that cannot meet their individual fuel consumption targets in the following year, require the manufacturer to submit an improvement plan (He & Yang, 2014a), and suspend the plant expansion application from noncompliant manufacturers. Such enforcement rules have not been updated for Phase IV standards.

The Chinese regulatory framework has a systemic flaw similar to the EU type-approval framework. Manufacturers pay laboratories to conduct tests, so there is a potential conflict of interest and it is difficult to ensure the test result is accurate. Moreover, regulators do not conduct regular independent testing to verify fuel consumption values of new vehicles and in-use vehicles, which increases the opportunities for manipulation and reporting unrealistic fuel consumption values.

## 2.6 JAPAN

#### 2.6.1 Vehicle testing

Japan has used a number of cycles to test vehicles for emissions and fuel economy. First, a test cycle called "10-mode" was used when emission standards for cars were first developed in 1973. The 10-mode cycle simulates urban driving conditions with a maximum speed of 40 km/h. The 10-15 mode, developed in 1991 and used to compile national fuel economy statistics from 1993, added one 15-mode segment simulating highway driving with speeds up to 70 km/h after three segments of the 10-mode. The JC08 test cycle was introduced in 2005 to determine vehicle fuel economy and was fully phased in by October 2011. Compared to 10-15 mode, JC08 is longer, has higher average and maximum speeds, and requires more aggressive acceleration. Moreover, fuel economy is measured twice, under cold-start and hot-start conditions. The final fuel economy value weighs the cold start by 25% and hot start by 75%. All tests are conducted at an ambient temperature of 20°C to 30°C. According to the Japanese government, the JC08 produces 9% lower fuel efficiency values compared with 10-15 mode (corresponding to a 10% increase in fuel consumption), which has made compliance with standards more difficult since the transition to the JC08.



Figure 12. Japanese driving schedules under 10-15 mode and JC08 (source: DieselNet, 2016)

Similar to the other countries, coastdown tests are used in Japan to determine road loads for chassis dynamometer testing. In 2016, Mitsubishi was found to have manipulated coastdown tests for decades. The manipulation led to as much as 5-10% higher fuel economy results than valid measurements (Soble, 2016).

# 2.6.2 Fuel economy and CO<sub>2</sub> emission targets

The Japanese government first established fuel economy standards for passenger vehicles in 1999 under its "Top Runner" energy efficiency program. Fuel economy targets are based on weight classes, with automakers allowed to accumulate credits in one weight class for use in another, subject to certain limitations. The 2010 target was 15.1 km/l (6.6 l/100 km or 35.5 mpg) under 10-15 mode and 13.6 km/l (7.4 l/100 km or 32 mpg) under JC08.

Because the majority of vehicles sold in Japan in 2002 already met or exceeded the 2010 standards, Japan revised its fuel economy standards in 2006, replacing the 10-15 mode cycle with the stricter JC08 test cycle and setting the 2015 target of 16.8 km/l (6.0 l/100 km or 40.0 mpg). In 2011, Japan released the 2020 standards of 20.3 km/l (4.9 l/100 km or 47.8 mpg) for passenger cars. On average, the new vehicle fleet exceeded its 2015 fuel economy target in 2011 and exceeded its 2020 fuel economy target in 2013.

A substantial share of the rapid fuel economy improvements in Japan can be attributed to the rise of hybrid electric vehicles (HEVs). In 2015, around 20% of passenger cars sold in Japan were HEVs, the highest share of any market. The aggressive fiscal incentives for fuel efficient vehicles have accelerated the growth of HEV shares since 2009 (Rutherford, 2015).



Figure 13. Japanese new vehicle fuel economy and regulatory targets, 1995 to 2020.

#### 2.6.3 Policy framework and enforcement

The Ministry of Economy, Trade and Industry (METI) and Ministry of Land, Infrastructure, Transport and Tourism (MLIT) jointly issue and enforce fuel economy standards.

For the type-approval application, all manufacturers must test the vehicles at National Traffic Safety and Environment Laboratory (NTSEL), the national laboratory owned and operated by government, or conduct testing with a witness from the NTSEL. While the testing is conducted or supervised by the governmental laboratory, manufacturers conduct coastdown tests. After the news broke about manipulation of coastdown tests by Mitsubishi, MLIT announced plans to conduct confirmatory road load tests or witness coastdown tests without previous notice.

For CoP, MLIT requires manufacturers to regularly test fuel economy values and conventional pollutant emissions of vehicles off the assembly line. MLIT investigates production lines to evaluate their capacity to produce qualified vehicles. No CoP testing is conducted by the regulatory agency, and MLIT does not require manufacturers to conduct in-use surveillance tests of any vehicles. MLIT selects some models to conduct in-use tests each year, but only to confirm compliance with conventional pollutant emission standards rather than fuel economy values.

There are minimal penalties if manufacturers fail to meet fuel economy targets. It is unclear how the regulatory agency will penalize manufacturers manipulating vehicles tests.

## 2.7 SUMMARY OF VEHICLE TESTING AND CO, STANDARDS

All four markets under study have fuel economy or  $CO_2$  standards in place (see Figure 14). The EU, United States, and China have similar targets in place, around 95 g  $CO_2$ /km (when normalized to NEDC), with the United States and China trailing behind the EU by five years. Japan and the EU currently have the most efficient new car fleets, with roughly 120 g  $CO_2$ /km according to official values.



Figure 14. Official passenger car CO<sub>2</sub> and fuel consumption values and targets, normalized to NEDC.

Table 2 compares the driving cycles used in the EU, the United States, China, and Japan. The CAFE procedure stands out with the highest mean velocity, lowest share of idling, and highest share of time accelerating. The Japanese JCO8 cycle has the lowest average speed and highest share of idling time and lies between the CAFE and NEDC schedules in terms of most measures of dynamicity. Lastly, the NEDC has the highest share of constant speed driving and a comparatively low average speed, but also has the highest mean positive acceleration.

A comparison of vehicle testing and policy frameworks in the four regions reveals substantial differences. The U.S. has the most extensive vehicle testing practices, with the U.S. EPA compliance program ensuring independent confirmatory tests, testing of vehicle configurations representing actual production, and in-use surveillance testing. In contrast, the other regions primarily rely on CoP to ensure that production vehicles meet type-approval specifications, although the EU's draft proposal to overhaul the type-approval framework would add several other testing components; in addition, introduction of the WLTP (applicable in the EU, Japan, and China) will tighten test conditions during laboratory testing. In terms of official CO<sub>2</sub> values, Japan has the most efficient fleet and the EU has the most ambitious targets. The rest of the study aims to assess how these differences in vehicle testing and policy frameworks affect the on-road efficiency of vehicles.

**Table 2.** Overview of test cycles used in the EU, the United States, China, and Japan (data source:Kühlwein et al., 2014).

	Units	NEDC	FTP75 weighted	HWFET	CAFE	JC08
Country		EU/China	US	US	US	Japan
Start condition		cold	43% cold / 57% hot	hot		25% cold / 75% hot
Duration	S	1180	1369	765		1204
Distance	km	11.03	11.99	16.51		8.17
Mean velocity	km/h	33.6	31.5	77.7	43.0	24.4
Max. velocity	km/h	120.0	91.2	96.4		81.6
Stop phases		14	18	2		12
		Dı	urations			
Stop	S	280	241	4		346
Constant driving	S	475	109	126		21
Acceleration	S	247	544	338		432
Deceleration	S	178	475	297		405
Shares (% of cycle duration)						
Stop		23.7%	17.6%	0.5%	13.3%	28.7%
Constant driving		40.3%	8.0%	16.5%	10.1%	1.7%
Acceleration		20.9%	39.7%	44.2%	40.8%	35.9%
Deceleration		15.1%	34.7%	38.8%	35.7%	33.6%
Mean positive acceleration	m/s²	0.59	0.50	0.19	0.42	0.42
Max. positive acceleration	m/s²	1.04	1.48	1.43		1.69
Mean positive velocity x acceleration, acceleration phases	m²/s³	4.97	3.86	3.45	3.76	3.34
Mean positive velocity x acceleration, whole cycle	m²/s³	1.04	1.53	1.52	1.53	1.20
Max. positive velocity x acceleration	m²/s³	9.22	19.19	15.17		11.60
Mean deceleration	m/s <sup>2</sup>	-0.82	-0.58	-0.22	-0.49	-0.45
Min. deceleration	m/s <sup>2</sup>	-1.39	-1.48	-1.48		-1.19

# 3. LITERATURE REVIEW

# 3.1 EU

Several studies have analyzed the growing divergence between type-approval and official CO<sub>2</sub> emissions in the European vehicle market. To arrive at an estimate of the gap, some of these studies have taken a top-down approach by aggregating large sets of in-use fuel consumption data from one or more EU member states. Other studies have followed a bottom-up approach and looked into the underlying causes of the gap in terms of both size and development over time.

The From Laboratory to Road series (Mock et al., 2014, 2012, Tietge et al., 2015, 2016) is the most extensive top-down analysis of the divergence between official and on-road  $CO_2$  emissions in the European market to date. The 2016 study aggregated 13 real-world fuel consumption data sources from seven European countries covering a total of 15 years (Tietge et al., 2016). Combining all data sources, the study found that the average  $CO_2$  gap increased from 9% in 2001 to 42% in 2015, and that the increase was particularly steep after 2008, when  $CO_2$  standards were first agreed on in the EU. Section 4.1 summarizes the findings.

Ligterink and Eijk (2014) analyzed real-world fuel consumption data from Dutch company cars covering model years 2004 to 2014.<sup>3</sup> The study found that the average divergence between on-road and official fuel consumption values increased from approximately 10% to 50% from 2004 to 2013. Ligterink and Smokers (2015) also conducted a study of Dutch company car data, focusing on the real-world performance of five top-selling plug-in hybrid electric vehicle models. The study found that drivers on average only covered about 30% of the type-approved electric mileage with electric drive on the road, and observed a decreasing trend in the electric drive share. The CO<sub>2</sub> gap from the PHEV sample increased from 169% to 176% from 2013 to 2015.

Stewart et al. (2015) built upon the top-down approach of Mock et al. (2014) and added a bottom-up analysis of the factors causing the  $CO_2$  gap. Both approaches found that the divergence increased from around 10% in 2002 to about 35% in 2014. The study concluded that the growth was mainly due to manufacturers exploiting flexibilities in testing procedures. Similarly, Kadijk et al. (2012) demonstrated that the application of test flexibilities in the EU increased significantly between 2002 and 2010. The study estimated that the use of flexibilities accounted for around 11 percentage points and 7 percentage points of the on-paper reduction in  $CO_2$  emissions achieved between 2002 and 2010 for passenger cars and LCVs respectively.

Mellios et al. (2011) and Ntziachristos et al. (2014) developed a model to predict on-road fuel consumption of a given vehicle model on the basis of type-approval fuel consumption, engine capacity, and vehicle mass. The studies stated that in-use fuel consumption of gasoline and diesel passenger cars tested between 2009 and 2011 was, on average, 11% and 16% higher than the official value. Other studies (e.g., Ligterink, Smokers, Spreen, Mock, & Tietge, 2016; Tietge, Mock, Franco, & Zacharof, 2017) propose similar models to predict real-world fuel consumption. The former study also provides evidence that the increase in the real-world gap after 2007 is linked to increased exploitation of flexibilities in the test procedure and increased use of technologies that provide disproportionate benefits under laboratory conditions.

<sup>3</sup> Company cars are a common job benefit in many European countries and represent roughly half of new car registrations in Europe.

# **3.2 UNITED STATES**

Since fuel economy standards were introduced in the United States in the mid-1970s, the discrepancy between laboratory and real-world fuel economy has been the subject of government studies, industry surveys, and scientific reports. Evidence of an on-road fuel economy shortfall led the U.S. EPA to adjust downward the fuel economy values reported to consumers starting in 1984 (Greene et al., 2015). The original adjustments were 10% for the city test and 22% for the highway test, for an overall adjustment of about 15%. Before introducing updated test methods for the U.S. EPA label values in 2008, the U.S. EPA analyzed several real-world data sources of fuel economy estimates (U.S. EPA, 2006). The studies reviewed provided a snapshot of the on-road fuel economy shortfall at the time, which was roughly 20%.

More recently, Greene et al. (2015) and previously Greene, Goeltz, Hopson, & Tworek (2007), conducted an in-depth statistical analysis of the discrepancy between official and in-use fuel economy based data from the MyMPG service, a fuel economy tracking service on FuelEconomy.gov. The most recent study analyzed a sample of 75,000 submissions and covered model years 1984 to 2015. The study furnishes evidence of a growing divergence between official and real-world  $CO_2$  emissions, but notes that the magnitude and direction of the gap was taken into account by the U.S. EPA. The study also pointed out that U.S. EPA window label values are a reasonable predictor of average on-road fuel economy, but that the considerable variance in on-road fuel economy experienced by different drivers detracts from the usefulness of the U.S. EPA label value in purchasing decisions.

## 3.3 CHINA

A number of studies investigated the divergence between real-world and type-approval CO<sub>2</sub> emissions in the Chinese vehicle market. Huo et al. (2011) found that the real-world fuel consumption of new cars sold in China in 2009 was about 16% higher than the type-approval value. The study was based on fuel consumption data from 153 vehicle models tracked by Chinese car owners on a local website. The study further used the International Vehicle Emission model to compare the type-approval driving cycle (NEDC) to several real-world city driving cycles and concluded that the NEDC failed to reflect city driving conditions, which was identified as one of the primary reasons for the gap.

Zhang et al. (2014) measured on-road  $CO_2$  emissions from 60 passenger cars in 2013 in Guangzhou, Beijing, and Macao using PEMS equipment. They found that on-road  $CO_2$  emissions normalized to the NEDC were 30% higher than type-approval figures. On-road measurements that were not normalized to the NEDC were 10% higher than the normalized values.

Lastly, Ding et al. (2015) and Qin et al. (2016) analyzed large samples of real-world fuel consumption data collected from XiaoXiongYouHao, a Chinese fuel tracking mobile application. The latter study found that the gap increased from 12% in 2008 to 27% in 2015 based on a set of more than 500,000 drivers' fuel consumption data. The studies investigate variations in the gap by vehicle segment, vehicle make, geographic region, and season.

# **3.4 JAPAN**

A handful of studies have raised concerns about the divergence between type-approval and real-world  $CO_2$  emissions from passenger cars in Japan. Kudoh (2012) observed an on-road fuel economy shortfall of 24% (32% fuel consumption gap) in e-nenpi.com data from fiscal years 2001 to 2004 and cites the inability of the 10-15 mode to reflect urban driving conditions as one of the main reasons for the divergence. Imported vehicles

were found to have a lower gap than domestic ones, suggesting that imported vehicles were not optimized for Japanese road conditions and thus had higher laboratory fuel consumption values. Schipper (2008) observed a decrease in Japanese official fuel consumption from 8.1 I/100 km in to 6.5 I/100 km in 2005 (according to 10-15 mode), while real-world values only appeared to decline from 11.3 I/100 km to 10.6 I/100 km in the same period. These values correspond to a 40% gap in 1995 and a 63% gap in 2005, although it should be noted that the real-world values are based on fuel sales rather than measurements from individual vehicles. This method of estimating real-world fuel consumption is prone to errors because it proves difficult to accurately measure the required inputs for the calculation (Schipper, Figueroa, Price, & Espey, 1993).

# 4. DATA ANALYSIS

## 4.1 EUROPE: MULTIPLE SOURCES

Data type	User-submitted, vehicle test, and fuel card on-road fuel consumption values
Data availability	Varies between data sources; approximately 90,000 vehicles per year
Data collection	Varies between data sources; fuel consumption data entered by vehicle drivers into a publicly available online database, recorded using a tank card when refueling at gas stations, or measured on the basis of test drives
Fleet structure, driving behavior	Varies between data sources: approximately 73% of vehicles were company cars, 26% private cars, and 1% vehicles selected for testing. Driving behavior varies.

#### Description

This section summarizes the real-world fuel consumption and  $CO_2$  data analyses conducted in the *From Laboratory to Road* studies focusing on Europe. A condensed version of the results published in the 2016 update of the series is presented here. The 2016 report was based on 13 data sources from seven European markets and included data on approximately 1 million private and company cars. Table 8 provides an overview of the samples used in the report.

The available data sources can be classified into the following three groups based on how data was collected: automatically recorded data from fuel card providers, user-submitted data from online fuel consumption tracking services, and data from on-road tests conducted by auto magazines and car clubs. Data sets from fuel card providers mostly include company cars, while user-submitted data predominantly cover private cars.

There are some sources of bias and transcription errors specific to each sample. Nevertheless, any biases are considered to be consistent over time and should not interfere with the observed trends. For a detailed description of each data source, see Tietge et al. (2016).

Country	Source	On-road data collection	Total vehicles	Vehicles per year	Mostly company cars	Data availability	Dating convention
	Spritmonitor.de	User-submitted; some information on driving style provided	134,463	~9,000		2001-2015	Build year
	LeasePlan	Fuel card system, fuel consumption automatically recorded	~180,000	~20,000	Х	2006-2015	Fleet year
Germany	AUTO BILD	Test route, fuel consumption measured before and after test drives	2,242	~280		2008-2015	Test date
	auto motor und sport	Test route, fuel consumption measured before and after test drives	1,885	~150		2003-2015	Test date

 Table 3. Summary of data sources used for the 2016 From Laboratory to Road report.

Country	Source	On-road data collection	Total vehicles	Vehicles per year	Mostly company cars	Data availability	Dating convention
	Allstar card	Recorded using fuel card	242,353	~24,000		2006-2015	Build year
United	honestjohn. co.uk	User-submitted; no details on driving style	97,291	~6,500		2001-2015	Model year
Kingdom	Emissions Analytics	Test route, Portable Emissions Measurement System testing	674	~170		2012-2015	Test date
Netherlands	Travelcard	Fuel card system, fuel consumption automatically recorded	275,764	~25,000	х	2004-2015	Build year
	Cleaner Car Contracts	Various data collection procedures	24,513	~3,500	Х	2010-2015	Fleet year
France	Fiches-Auto.fr	User-submitted	23,559	~1,500		2001-2015	Model year
Spain	km77.com	Test route, fuel consumption measured before and after test drives	273	~45		2010-2015	Test date
Sweden	auto motor & sport	Test route, fuel consumption measured before and after test drives	643	~90		2009-2015	Test date
Switzerland	Touring Club Switzerland	Test route, fuel consumption measured before and after test drives	271	~20		1996-2015	Test date
Total			~1,000,000	~90,000			-

## Methodology

Thirteen data sets were used to estimate the divergence between real-world and typeapproval  $CO_2$  emissions in the European market. The analysis of each of sample followed the methodology described in Section 1. A description of the exact methodology used for each data set can be found in Tietge et al. (2016).

A central estimate of the divergence between European type-approval and real-world  $CO_2$  values was constructed by combining the 13 data sets. This process involved calculating the annual average divergence from all private car data sources. The same procedure was applied to company car data sets. Private and company car estimates were then combined with equal weights under the assumption that the European new car market consists of private and company cars in equal parts (Næss-Schmidt & Winiarczyk, 2010).

#### Results

Figure 15 shows the trend in the divergence between real-world and official  $CO_2$  emission values for each European data source. All data sources— regardless of country, data collection methodology, or whether cars are privately or company owned—show a clear upward trend in the divergence over time. Estimates of the divergence were approximately 9% in 2001, but increased to between 35% and 61% in 2015.

The precise level of the gap varies by data source. Vehicle tests by car magazines and other organizations typically produce internally consistent data due to standardized measurement procedures, but different sources deliver quite dissimilar results due to differing test procedures and small data sets. In addition, despite precise data collection procedures, fluctuations in traffic and weather conditions may affect the consistency of the tests. These inconsistencies are considered to be of minor import when looking at trends. The three data sources that rely on user input produce comparable divergence estimates over the years despite covering different markets, namely Germany (Spritmonitor.de), the UK (honestjohn.co.uk), and France (Fiches-Auto.fr). Estimates of the divergence from web services range from 8% to 11% in 2001 to between 35% and 42% in 2015. Divergence estimates of company cars are typically among the highest because company car drivers have a weaker incentive to drive in a fuel-conserving manner, as employers usually cover fuel expenses. Estimates in 2015 ranged from 41% to 61% in 2015.



Build year / Heet year / Hodel year / Test year

**Figure 15.** Divergence between type-approval and real-world CO<sub>2</sub> emissions for various European on-road data sources.

Figure 16 combines divergence estimates from different data sources and shows averages for private cars, company cars, and the total new vehicle market. Combining the 13 data sources, the divergence increased from 9% in 2001 to 42% in 2015. Over that time, company cars consistently exhibited a higher average divergence than private cars, as noted above. In 2015, the average estimate of the divergence from company cars amounted to 45%, five percentage points higher than the private car average.

Considering that the data sources cover different European markets, include private and company cars, and follow differing data collection procedures, the central estimate should be viewed as strong evidence of a systemic, growing divergence between real-world and official CO<sub>2</sub> emission values. Estimates of the divergence vary from source to source, but aggregating data on almost 1 million cars reveals a consistent increase in the divergence over time that seems to affect all European markets alike.





# 4.2 UNITED STATES: FUELECONOMY.GOV

Data source	https://www.fueleconomy.gov/mpg/MPG.do
Data type	On-road, user-submitted
Data availability	Model years 2001–2014, approximately 3,000 vehicles per model year
Data collection	Fuel economy data entered by vehicle drivers into a publicly available online database
Fleet structure, driving behavior	City and highway driving, some information on driving conditions and fuel costs

#### Description

FuelEconomy.gov<sup>4</sup> is the official U.S. government online source for fuel economy information. The site is maintained by the U.S. Department of Energy with data provided by the U.S. EPA. FuelEconomy.gov helps consumers make informed fuel economy choices when purchasing vehicles and advises them on how to drive more efficiently. The website provides EPA fuel economy ratings for passenger cars and trucks, user-provided real-world fuel economy estimates, and extensive information on current fuel and technology trends. Real-world fuel economy estimates are collected through MyMPG, a tool embedded in FuelEconomy.gov that allows users to track their fuel economy and compare it with EPA label values and input from other drivers. The service currently has more than 100,000 active users.

MyMPG enables users to track their fuel economy values using one of two methods. They can enter fuel log data, including odometer readings and fuel quantity, and MyMPG automatically calculates the fuel economy for a driving period. Alternatively, they can directly enter fuel economy estimates calculated by them or obtained from the vehicle's digital dash readout. For the first method, users are requested to fill the vehicle's gas tank completely and write down odometer readings each time they refuel the car. Once two readings are entered, the tool can calculate fuel economy. Users can also add information on fuel cost and driving conditions to each record. Driving conditions are reported as the shares of city and highway driving.

## Methodology

The FuelEconomy.gov sample initially consisted of two different real-world fuel economy data sets, reflecting the method chosen by the user to enter the data. The first group consisted of fuel log data from more than 37,000 vehicles. The second group contained fuel economy estimates calculated by users for more than 46,000 vehicles. Fuel consumption values from the first group were estimated by evaluating the longest sequence of valid fuel-up data for each vehicle. After calculating fuel consumption estimates for the first group and converting fuel economy values from the second group to fuel consumption values, both data sets were combined.

For each vehicle in the combined data set, the divergence between real-world and official fuel consumption was calculated using both the CAFE and U.S. EPA label values as denominators. The U.S. EPA changed the methodology used to estimate fuel economy starting with MY 2008 (see Section 2.4), so divergence values for both types of label values are presented in the analysis. Extreme divergence estimates resulting from transcription errors were removed using Peirce's criterion.<sup>5</sup> In total, 43,000 out of 83,000 vehicles remained after removing approximately 14,000 vehicles with invalid fuel log data, roughly 3,000 vehicles with missing fuel economy values, approximately

<sup>4</sup> See http://fueleconomy.gov. The complete data set was generously provided by Rick Goeltz and Janet Hopson of the Oak Ridge National Laboratory.

<sup>5</sup> For more information on Peirce's criterion, see Tietge, Mock, Franco, & Zacharof (2017).

20,000 vehicles from before model year 2001, and less than 1,000 vehicles with unrealistic divergence estimates.

Because MyMPG users enter fuel consumption data on a voluntary basis, there is a risk of self-selection bias in the data (although any bias should be reasonably consistent over time, so trends should be reasonably accurate). For instance, the service may attract consumers who are particularly concerned about fuel economy. Greene et al. (2015) analyzed the MyMPG data as of mid-2015 and compare MyMPG vehicles to new cars registered in the United States. The comparison provides a rough indication of the sample representativeness. Table 4 presents Greene et al.'s (2015) findings together with a summary of the MyMPG sample at hand. As illustrated in the table, passenger cars (excluding SUVs) appear to be overrepresented in the MyMPG sample, while the remaining vehicle types are somewhat underrepresented.

	New vehicles as a share of production MY 2000-2014			MyMPG (Greene et al., 2015)	MyMPG (this study)		
				MY 2000-2014	MY 2001-2014		
Vehicle type	Mean Min Max		Mean	Mean			
Car	52.70%	47.70%	60.50%	67.56%	64.48%		
SUV	27.70%	18.90%	17.72%	17.72%	22.93%		
Pickup	13.40% 10.10% 10.42%		10.42%	8.39%			
Minivan/van	6.20% 3.80% 10.20%		10.20%	4.30%	4.20%		

Table 4. Comparison of MyMPG shares by vehicle type to market shares (based on: Greene et al., 2015).

In addition to the comparison based on vehicle types, average vehicle specifications of MyMPG cars are compared with sales data for model year 2014 in Table 5. While the average capacity and number of cylinders of the MyMPG sample are in line with the market average values, diesel and hybrid cars are significantly overrepresented. As a result, average CO<sub>2</sub> emissions of the MyMPG data set are roughly 20% lower than the average of new 2014 cars. These figures indicate that diesel vehicle and HEV owners are more likely to use a web service such as MyMPG.

**Table 5.** Comparison of average characteristics of MyMPG and new passenger cars in model year2014 (source of sales data: U.S. EPA, 2015b).

Source	Model year	Capacity (l)	No. cylinders	Diesel (%)	Automatic transmission (%)	HEV (%)	CO <sub>2</sub> (g CO <sub>2</sub> /km)
MyMPG	2014	2.2	4.3	8.0%	86%	22.1%	136
Sales	2014	2.4	4.5	1.3%	96%	4.2%	167

#### Results

Figure 17 shows the trend in the divergence between real-world and official CO<sub>2</sub> emission values based on CAFE and U.S. EPA label values.<sup>6</sup> Taking the former as a reference, the divergence increased from around 16% in MY 2001 to 34% in MY 2015 for all light-duty vehicles, and from 14% to 31% for cars only. U.S. EPA label values more accurately reflect real-world fuel consumption figures, but the divergence between U.S. EPA and real-world fuel consumption also increased over time. Before MY 2008, U.S. EPA label values were calculated using a flat adjustment factor for the city and highway fuel economy values (see Section 2.4). The divergence between real-world fuel consumption and the U.S. EPA label values therefore consistently lies about 17 percentage points below the CAFE divergence, increasing from roughly -1% in MY 2001 to 4% in MY 2007. The new U.S. EPA label methodology, the five-cycle method, produces the most realistic fuel consumption values in recent years, with divergence estimates ranging from approximately -8% in MY 2008 to close to no divergence MY 2015.

<sup>6</sup> Results are presented in terms of fuel economy in Appendix I.



**Figure 17.** Divergence between MyMPG and official CO<sub>2</sub> emission values for light-duty vehicles and cars. Official CO<sub>2</sub> values include CAFE figures and EPA label values.

Figure 18 shows the development of the divergence between real-world and CAFE values for different powertrains.<sup>7</sup> The lines represent the development of the real-world divergence for cars including and excluding hybrid electric vehicles (HEVs). HEVs consistently have higher average divergence values than conventional powertrains, as witnessed by the higher divergence for both CAFE and U.S. EPA values when including HEVs. For CAFE values, the difference between including and excluding HEVs reached four percentage points in model year 2015, when HEVs accounted for approximately 11% of the cars in the data set. Diesel cars were not plotted separately due to the low share (less than 5% of cars in the sample) and the likelihood of overstated real-world fuel economy due to the VW defeat devices. The results show that the 2008 fuel economy label method significantly reduced the gap for HEVs compared with other vehicles.



**Figure 18.** Divergence of MyMPG from official CO<sub>2</sub> emission values by powertrain. Official CO<sub>2</sub> values include CAFE figures and EPA label values.

<sup>7</sup> Results are presented in terms of fuel economy in Appendix I.

Figure 19 shows the average divergence between FuelEconomy.gov and CAFE fuel consumption values by data collection method. The first two bars refer to fuel log data, one based on cumulative odometer readings ("odometer"), the other based on incremental distance readings ("trip"). The last two bars show the divergence according to estimates entered directly by users, including fuel consumption values measured by the vehicle's onboard computer ("digital readout") and fuel consumption values calculated by the user ("mathematical"). The figure shows that fuel log data generally provides higher divergence estimates than when users directly input real-world fuel estimates.



Figure 19. Divergence between MyMPG and CAFE CO., emission values by data entry method.

# 4.3 CHINA: XIAOXIONGYOUHAO

Data source	http://www.xiaoxiongyouhao.com/
Data type	On-road, user-submitted
Data availability	2005-2015, approximately 50,000 vehicles per model year
Data collection	Fuel consumption data entered by vehicle drivers into an online database using a mobile app
Fleet structure, driving behavior	Urban and extra-urban driving

#### Description

XiaoXiongYouHao<sup>8</sup> is a Chinese mobile application that allows users to track and compare their fuel consumption. The app was launched in 2010 and has more than 2.5 million downloads in China. XiaoXiongYouHao publishes periodic reports analyzing the real-world fuel consumption data to help policy makers and consumers make better-informed decisions.

To start using the XiaoXiongYouHao application, drivers are requested to select their vehicle model version. Users then log fuel volume and odometer readings and indicate whether the tank is full after refueling. The application calculates fuel consumption based on the reported fuel volume and vehicle mileage once sufficient data has been entered. A typical user records more than 20 fueling events.

#### Methodology

The XiaoXiongYouHao data set included information on approximately 7,000 vehicle model variants with model years ranging from 2003 to 2016. More data points were available for recent years because the Chinese markets for both vehicles and smartphones grew during the measured time frame. For each vehicle model variant, the sample included information on the model year, engine specifications, transmission type, vehicle segment, average on-road and official fuel consumption, and number of vehicles. After filtering the data set for missing values and removing model years 2003 and 2004 as they included fewer than 500 vehicles per year, a sample of about 490,000 vehicles remained, with model years ranging from 2005 to 2015. Outlier removal was not deemed necessary as no erroneous data points were identified using Peirce's criterion.

Because XiaoXiongYouHao users record fuel consumption data on a voluntary basis, there is a risk of self-selection bias. To assess the sample representativeness, Table 6 compares average fleet parameters with the corresponding average values of newly registered passenger vehicles in China in 2014. In addition, Figure 20 compares the XiaoXiongYouHao sample with the 2014 Chinese new car fleet in terms of segment shares. Since the XiaoXiongYouHao data set uses model year, and market data was only available by registration year, 2014 sales data were compared to XiaoXiongYouHao segment shares and fleet specifications averaged over model years 2011 to 2014, under the premise that new registrations mainly consist of vehicle models released during the four years preceding the date of purchase.

As illustrated in Table 6, the average engine power, engine capacity, and type-approval fuel consumption values of the XiaoXiongYouHao sample match the 2014 new fleet average values very closely. The share of automatic vehicles in the sample is slightly higher than the 2014 new fleet share, whereas the opposite is true for the average vehicle mass. The discrepancy in the mass values is consistent with the

<sup>8</sup> See http://www.xiaoxiongyouhao.com. The complete data set was generously provided by the data proprietor.

overrepresentation of small and lower medium cars in the XiaoXiongYouHao sample, as shown in Figure 20. The XiaoXiongYouHao sample is generally somewhat skewed toward smaller vehicles, with medium and MPV segments underrepresented. Despite these deviations, on the whole the sample appears to provide a reasonable reflection of the Chinese new vehicle market.

**Table 6.** Comparison of vehicle characteristics in the XiaoXiongYouHao sample (model years 2011 to 2014) and new passenger vehicles for sales year 2014 (source: data provided by Segment Y).

Source	Year	Engine power (kW)	Engine displacement (l)	Fuel consumption (I/100km)	Curb weight (kg)	Automatic transmission (%)
XiaoXiongYouHao	Model years 2011-2014	99	1.7	7.1	1343	60%
Sales data	Sales year 2014	101	1.7	7.3	1376	56%



**Figure 20.** Segment shares in the XiaoXiongYouHao sample (model years 2011-2014) and Chinese vehicle sales for sales year 2014 (source: data provided by Segment y).

Another potential bias should be acknowledged: XiaoXiongYouHao users enter data using smartphones. Use of smartphone apps for fuel expense recording may be more prevalent in urban areas, leading to an overrepresentation of city driving in the XiaoXiongYouHao sample and potential overestimation of real-world fuel consumption. Nevertheless, if such a bias were present in the data, we would expect more pronounced differences between average vehicle characteristics in the XiaoXiongYouHao data and official registration statistics, assuming that average vehicle characteristics differ depending on where they are purchased.

#### Results

Figure 21 plots the development in the average divergence between XiaoXiongYouHao and official  $CO_2$  emission values by transmission type. Model years with less than 5,000 vehicles are plotted using sparsely dotted lines.

According to the XiaoXiongYouHao sample, the average divergence between real-world and type-approval  $CO_2$  emissions increased by around 15 percentage points between MY 2008 and MY 2015, reaching 27% in model year 2015. As Figure 21 shows, the divergence increased for both automatic and manual transmission cars, and vehicles with automatic transmissions consistently exhibited a higher divergence than vehicles with manual transmissions. Until MY 2013, the shares of automatic and manual transmission vehicles were balanced. From MY 2014 on, automatic vehicles clearly outnumbered those with manual transmission, and in MY 2015 they made up 67% of the XiaoXiongYouHao fleet.



**Figure 21.** Divergence between XiaoXiongYouHao and official CO<sub>2</sub> emissions by transmission type (pie chart indicates the share of vehicles per transmission type in the data set for model year 2015).

## 4.4 JAPAN: E-NENPI.COM

Data source	www.e-nenpi.com
Data type	On-road, user-submitted
Data availability	Data from 2001 to 2014, approximately 3,000 vehicles per year
Data collection	Fuel economy data entered by vehicle drivers into a publicly available online database
Fleet structure, driving behavior	Unknown

#### Description

Data from the Japanese web service e-nenpi.com was used to estimate the gap between official and real-world fuel consumption values in Japan. The web service allows users to monitor their fuel consumption by entering fuel quantity and odometer readings. More than 80,000 vehicles have been added to e-nenpi.com.

#### Methodology

The e-nenpi.com sample included approximately 20,000 observations. Each observation contains one year of data for a given model version.<sup>9</sup> Because different vehicles of a model version may have entered and left the database over time, and data was delivered aggregated by model version and calendar year, the number of vehicles in the sample was estimated. In total, more than 47,000 cars from model years 2001 to 2014 were analyzed. Since data was delivered in aggregated form, outlier detection was not performed for individual vehicles, but data for vehicle models were validated using Peirce's criterion. A detailed methodology is provided in Appendix II.

To evaluate the representativeness of the e-nenpi.com sample, segment shares of the e-nenpi.com fleet were compared with those of the 2014 new car fleet in Japan. In the absence of market data by model year, which is the dating convention of the e-nenpi.com sample, it was assumed that the new fleet is comparable to e-nenpi. com segment shares averaged over model years 2011 to 2014. In addition, average vehicle specifications of Japan's 2011 passenger vehicle fleet were compared to the specifications provided in the e-nenpi.com data set averaged over model years 2008 to 2011. A comparison for more recent years was not possible due to data availability. Segment information was not provided in the e-nenpi.com data set. Vehicle model names were thus translated from Japanese to English, and the data set was joined with external data that contained the required segment information.

Figure 22 shows that segment shares of new 2014 vehicles decrease as vehicle size increases. The mini segment stands out with a 40% share of new registrations. The mini segment consists mainly of Kei cars, a category specific to the Japanese vehicle market that is strongly incentivized by the government (German, 2015). In the e-nenpi.com data set, however, the share of mini cars only amounts to roughly 20%. While the sample shares of small, upper medium, and large vehicles match with the respective shares of new vehicles sold in 2014, the lower medium and medium segments are significantly overrepresented. These findings are in line with the comparison presented in Table 7. The average capacity and HEV share of the e-nenpi.com data set are higher than those of the 2011 fleet. The share of HEVs in the Japanese market started increasing in 2009 (Rutherford, 2014), and the most popular HEV models, such as the Toyota Prius, belong to the lower medium and medium segments.

<sup>9</sup> A model version was defined as a unique combination of vehicle brand, model name, model year, and a number of powertrain characteristics.





**Figure 22.** Share of new passenger vehicles in Japan in 2014 by segment vs. segment shares in the e-nenpi data set for model years 2011 through 2014.

In short, the e-nenpi.com sample is skewed towards mid-sized cars, whereas Japan's unique Kei segment is significantly underrepresented. In addition, the size of the sample is relatively small, which further undermines the statistical significance of the analysis. Nevertheless, this study is among the first to tackle the divergence between official and real-world  $CO_2$  emissions in the Japanese vehicle market and serves as a basis for subsequent research.

#### Results

Figure 23 presents the development of the average divergence between official and e-nenpi.com fuel consumption values over vehicle model years.<sup>10</sup> While the divergence between 10-15 mode values and e-nenpi.com data is more or less constant during the model years 2001 through 2007, there is a noticeable increase after 2007. The increase is also noticeable in the JC08 data. This development followed the introduction of the 16.8 l/km (5.9 l/100 km) fuel economy target for 2015, which was agreed on in 2007, and subsidies for vehicles exceeding the 2010 fuel economy standard by 15% or more, which were introduced in 2009 (see Section 2.6 for details on fuel economy targets).

<sup>10</sup> Results are presented in terms of fuel economy in Appendix I.



**Figure 23.** Divergence between official and e-nenpi.com CO<sub>2</sub> emission values for different vehicle model years.

Figure 24 compares the average divergence between e-nenpi.com values and the two test cycles, the 10-15 mode and the JC08, and distinguishes between conventional powertrains and hybrid electric vehicles. Only vehicles with both 10-15 mode and JC08 fuel consumption values were included in the graph, the vast majority of which were of model years 2008 to 2011, so the difference between the different gap estimates is directly attributable to the differences between the two test cycles. The figure illustrates that the 10-15 mode generally delivered less realistic fuel consumption values: for conventional powertrains, the divergence was almost nine percentage points higher using the 10-15 mode. For HEVs, this difference is even more pronounced, with the 10-15 mode gap exceeding the JC08 gap by approximately 20 percentage points.

![](_page_46_Figure_4.jpeg)

Figure 24. Divergence between official and e-nenpi.com CO, emission values for different powertrains.

# 4.5 DATA COMPARISON

Table 8 provides an overview of the data sources used in the analysis. In total, the analysis covers more than 1.5 million vehicles from 2001 to 2014.

Region	Data source	Total vehicles	Vehicles per year	Dating convention
China	XiaoXiongYouHao	~490,000	~50,000	Model year
EU	13 data sources	~1,000,000	~90,000	Various
Japan	e-nenpi.com	~47,000	~3,000	Model year
United States	FuelEconomy.gov	~41,000	~3,000	Model year
Total	-	~1.6M	~146,000	-

Table 8. Summary of data sources used in the analysis.

Figure 25 plots the divergence between real-world and official  $CO_2$  emission values of passenger cars in the four regions covered in the analysis, revealing substantial differences between the markets. According to e-nenpi.com data, Japan is the market with the highest divergence. In contrast, EPA label values provide the most realistic  $CO_2$  values. Between these two extremes, U.S. CAFE and EU and China NEDC values all exhibit growing gaps between declared and on-road  $CO_2$  emission values, which increase from an 8% to 14% range in 2001 to a 25% to 40% range in 2014. The growth in the gap is considerably less pronounced in the United States than in the EU and in China.

![](_page_47_Figure_6.jpeg)

**Figure 25.** Divergence between official and real-world CO<sub>2</sub> emissions for new passenger cars in the EU, the United States, China, and Japan.

Figure 26 plots the 2014 estimates of the gap. The estimates for the four regions are the same as the 2014 values in Figure 25, but are also presented in fuel economy terms. As fuel consumption and fuel economy are inversely related, the curve representing the relationship between the fuel consumption gap and fuel economy shortfall is convex to the origin.

![](_page_48_Figure_1.jpeg)

**Figure 26.** 2014 estimates of the divergence between official and real-world fuel consumption and fuel economy values in different regions.

Figure 27 focuses on the increase in the gap and plots the compound annual growth rate (CAGR) for different regions and time periods. Between 2001 and 2014, the annual growth rate in the EU (13% CAGR) is roughly twice as high the one observed in the U.S. market (7% CAGR). The gap in Japan is the most stable when comparing 2001 and 2014 values (2% CAGR). This metric is not available for China because on-road data only reaches back to 2005.

Focusing on the 2008 to 2014 time frame reveals additional information. China has the highest annual growth in divergence, followed by the EU. Both markets use test procedures based on the NEDC. The EU also exhibits a noticeably higher annual growth rate during the 2008 to 2014 time frame than during the 2001 to 2014 period. This change indicates that the gap grew quicker after the 2008 agreement on the first European  $CO_2$  standards. In contrast, the United States had slightly lower annual growth rates for 2008-2014. As the United States had flat standards for cars from 2001 to 2010 and increasing standards starting in 2011, this means that the gap increased faster when standards were not changing than when they did change—exactly the opposite from Europe. The Japanese JC08 values show a substantial increase between 2008, the year the new test procedure was introduced, and 2014.

![](_page_48_Figure_5.jpeg)

**Figure 27.** Annual growth rates (CAGR) of the divergence between official and real-world  $CO_2$  emissions for various regions (expressed as % growth of gap).

# 5. DISCUSSION

#### 5.1 COMPARISON OF RESULTS AND PREVIOUS RESEARCH

This study is not the first attempt to analyze on-road fuel consumption data from the foregoing four vehicle markets. The results for Europe were extracted from previous From Laboratory to Road studies by the ICCT (e.g., Tietge et al., 2016), and the analyses for the other regions followed in other researchers' footsteps. For the United States, our results closely match findings by Greene et al. (2015), who also analyzed the MyMPG data from FuelEconomy.gov. Both the level and pattern in the CAFE gap appear congruent. For China, Ding et al. (2015) also analyzed data from XiaoXiongYouHao and gap estimates closely match our results, with the largest difference between annual estimates being three percentage points. Lastly, there is less research to compare to for Japan, but our findings are similar to results from Kudoh (2012), with both studies pointing to a gap of roughly 30% for the 2001 to 2004 time frame. Our results do not match the dramatic increase in the gap, from 40% in 1995 to 63% in 2005, that Schipper (2008) estimated, although it should be noted that these numbers are based on market-wide fuel sales rather than measuring the fuel consumption of individual vehicles. The former method is susceptible to systemic errors in the input data (Schipper et al., 1993). Nevertheless, more research and data is needed to establish reliable real-world fuel consumption values for the Japanese market. Taken together, the findings presented in this study are generally in line with previous research, indicating that the methodological approach and data used in this analysis are sound and confirming results from preceding studies.

## 5.2 REASONS FOR GROWING DIVERGENCE ESTIMATES

A detailed discussion of the underlying reasons for the growing divergence between official and on-road  $CO_2$  emission values is outside the scope of the study, but some common reasons are discussed below.

## 5.2.1 Decreasing official CO<sub>2</sub> values

A frequently cited reason for the increasing gap is the global decrease in official  $CO_2$  values, which makes a constant absolute g/km difference between real-world and type-approval values proportionately larger.<sup>11</sup> This factor was determined to be of minor import in the EU, where the decrease in type-approval  $CO_2$  values accounts for less than one-tenth of the increase in the gap between 2001 and 2015 (see Tietge et al., 2016, section 4). Replicating this calculation for the United States indicates that roughly one-fifth of the increase in the gap between model years 2001 and 2014 may be due to declining  $CO_2$  values.<sup>12</sup> In China, this effect may explain roughly one-tenth of the increase in the gap of the drivers of the growing gap: The absolute gap in Japan decreased from 2001 to 2014.<sup>14</sup> This development is mainly due to the fact that Japan had a comparatively high divergence to begin with (37%), and saw a drop of more than 40% in official fuel consumption values from 2001 to 2014.

<sup>11</sup> In reality, while there may be a component of the gap that has a constant absolute g/km difference, most of the gap is proportional. Thus, this methodology overstates the impact of decreasing official  $CO_2$  values on the gap.

<sup>12</sup> According to NHTSA data (U.S. EPA, 2015b), average fuel consumption of U.S. cars decreased from 8.2 I/100 km (28.8 mpg) in model year 2001 to 6.5 I/100 km (36.4 mpg) in MY 2014. The absolute gap in MY 2001 was 14% × 8.2 I/100 km ≈ 1.1 I/100 km according to MyMPG data. If this absolute offset had remained constant, the divergence in MY 2014 would have been roughly 17%. Instead, the observed gap that year was 31%.

<sup>13</sup> Average official  $CO_2$  values of new passenger cars in China decreased from 185 to 169 g  $CO_2$ /km from 2008 to 2014. At the same time, the gap increased from 11% to 25%. The 2014 gap would have been 12% if the absolute offset had remained stable since 2008.

<sup>14</sup> Average fuel consumption of cars in Japan decreased from 7.9 I/100 km in 2001 to 4.6 I/100 km in 2014, where both values are normalized to the JC08. According to e-nenpi.com data, the absolute gap in 2001 was 37% × 7.9 I/100 km ≈ 2.9 I/100 km. If this gap had remained constant until 2014, the observed divergence in MY 2014 would have been higher than the observed gap of 46% (or 2.1 I/100 km in absolute terms).

of the limited Japanese real-world data at hand, more study of this development is recommended. On the whole, while these calculations do not provide definitive evidence of the influence of decreasing official  $CO_2$  values on the gap, they indicate that this effect is not the predominant driver of the growing divergence in the EU, the United States, and China, although it may be at least a contributing factor, especially in Japan.

#### 5.2.2 Driver behavior

Driver behavior is also cited as a reason for the growing gap in some markets. Selfreported data on driving behavior is available for two data sources in this study, namely Sprimonitor.de (EU) and FuelEconomy.gov (United States), and both data sets confirm that driving behavior affects on-road fuel efficiency. The Spritmonitor.de data indicates that economical driving reduces the gap by approximately nine percentage points compared with balanced driving, while speedy driving increases the gap by 7 percentage points compared with a balanced driving style. However, different driving styles did not account for the increase in the gap over time, as the share of different driving styles remained fairly constant and all driver types experienced an increase in gap estimates over time (Tietge et al., 2016). An extensive regression model developed by Greene et al. (2015) indicates that driving style significantly affected on-road fuel economy. For gasoline vehicles, cautious drivers improved their fuel economy by 9% to 10% (8% to 9% decrease in fuel consumption) compared with the most aggressive drivers. More importantly, however, the shares of self-reported driving styles remained fairly constant over time (see Figure 28). While it is possible that other developments (e.g., increased speed limits, increased opportunity costs of driving, increased vehicle performance, etc.) may affect all driving styles alike, the self-reported data on FuelEconomy.gov indicates that driving behavior does not account for the increasing gap. Taken together, self-reported data on driving behavior from the EU and United States suggest that changes in driving styles do not account for the growing gap in the two regions.

![](_page_50_Figure_4.jpeg)

Figure 28. Shares of different driving styles in MyMPG data.

#### 5.2.3 Vehicle technologies

A number of recent developments in vehicle technologies contribute to the growing gap. For instance, air-conditioning systems have become more common in all markets

included in the analysis but are turned off during laboratory tests; the only exception to this rule is the SC03 cycle, which forms part of the U.S. EPA five-cycle method. Other vehicle technologies, such as stop/start systems, have been shown to typically provide disproportionate benefits during laboratory testing compared with real-world driving in Europe and China (Qin et al., 2016; Stewart et al., 2015), while hybrid electric vehicles show higher gaps in the EU, United States, and Japan (data for China was not available).

Increased ethanol content of fuels also contributes to the gap since ethanol has a lower energy content on a volumetric basis, so real-world fuel consumption could be inflated if official fuel consumption values fail to consider the increase in ethanol content over time. In the United States, the increase in ethanol accounts for around two percentage points of the divergence between real-world and official CO<sub>2</sub> values (Greene et al., 2015) and three percentage points of the gap between CAFE and U.S. EPA five-cycle values (U.S. EPA et al., 2016). In the EU, biofuels have become more common over time. Ethanol blends E5 and E10 (5% and 10% ethanol share respectively) made up more than 80% of all gasoline sold in 2014, while B7 (7% biodiesel share) accounted for virtually all diesel sold in the same year (EEA, 2015); however, these developments are largely taken into account in the calculation of type-approval fuel consumption figures by means of reference fuels (see UNECE, 2013) and should therefore only increase the gap of older vehicles, for which official fuel consumption figures do not consider biofuel blends.

#### 5.2.4 Vehicle testing and policy frameworks

To what extent manufacturers are optimizing vehicles for testing in order to produce lower  $CO_2$  emission values is a key question for policymakers. This analysis does not attempt to answer that question, but we discuss findings from other studies relative to our results. Since testing procedures and policy frameworks vary by jurisdictions, findings are organized by region.

In Europe, numerous studies indicate that test cycle optimization and the exploitation of flexibilities in the test procedure contribute to—and may be the leading cause of—the growing gap. A detailed summary of findings is beyond the scope of the discussion, but unrealistic road-load coefficients from coastdown testing and leveraging flexibilities and tolerances in chassis dynamometer testing are key culprits (Kadijk et al., 2012; Mellios et al., 2011; Stewart et al., 2015). The European policy framework perpetuates these problems because market surveillance is limited and road-load coefficients are treated confidentially and are not verified by regulators. Moreover, the European Commission currently does not have the power to issue vehicle recalls or impose financial penalties in case of violations.

In contrast to the EU, there is no clear evidence to link the increase in the U.S. CAFE gap to manufacturers optimizing test values. Not only is the increase in the gap considerably lower than in the EU (see Figure 27), but the annual growth in the gap actually slowed somewhat after the United States adopted new standards for cars starting in 2011, indicating that the U.S. policies provide less room for manipulation. Greene et al. (2015) observe that "the timing of the increase in [the U.S. on-road fuel economy] shortfall suggests the possibility that as standards are tightened (or gasoline prices increase) manufacturers increasingly design vehicles to perform well on the test cycle [...], in the process increasing the gap between test cycle and on-road fuel economy" (p. 22). However, the study concludes that more research is needed to establish causality. The regression model presented in the same study also indicates that individual model year effects rarely have significant impacts on real-world fuel economy when accounting for a range of vehicle and external parameters, but that "the pattern of the model year coefficients [...] suggests the possibility of an increasing shortfall in the more recent model years when fuel economy standards have been increasing" (Greene et al., 2015, p. 26). The U.S. policy framework incorporates a wide range of measures that aim to detect misstated fuel economy values, which may explain the lower growth in the gap compared to other markets. These measures include SEAs, road-load confirmatory testing, and in-use surveillance testing (see Section 2.4.3), which have led to numerous corrections of fuel economy values (U.S. EPA, 2014a). Comparing EU and U.S. road load coefficients, Kühlwein (2016) notes that U.S. coefficients reported by manufacturers are far closer to independent measurements and traces this difference back to rigorous policy enforcement in the United States rather than better coastdown procedures. Fines for violations act as a further deterrent. For instance, Hyundai and Kia paid a civil penalty of \$100 million for using inaccurate road load forces, thereby overstating fuel economy of their vehicles (U.S. EPA, 2014b). Taken together, there is no clear evidence linking the increase in the U.S. gap to manufacturers gaming the system, and numerous mechanisms are in place to detect any violations and to deter manufacturers from cheating. This is supported by the fact that the annual increase in the gap in the United States actually slowed down slightly after new car standards were adopted for 2011 through 2016.

There is less literature investigating the growing gap in China and Japan. Chinese studies seem to indicate that the test procedure rather than the test cycle itself is the leading cause of the divergence between official and real-world  $CO_2$  emission values, given that on-road emissions were 30% higher than laboratory measurements, even when the real-world measurements were normalized to the NEDC (Zhang et al., 2014). Another factor that was highlighted by previous studies is the importance of traffic conditions, and particularly traffic congestion, on real-world  $CO_2$  emissions (Qin et al., 2016; Zhang et al., 2014). Similar to findings on the gap in China, Kudoh (2012) links the divergence in Japan to the inability of the 10-15 mode to accurately recreate congested city driving conditions.

#### 5.2.5 Summary

The reasons for the growth of the divergence in several regions of the world are likely to be a complex combination of different developments; however, literature from the EU and China indicates that the exploitation of loopholes in the NEDC is the leading cause. In contrast, there is no clear evidence of systematic manipulation in the United States or Japan.

## 5.3 POLICY IMPLICATIONS

The divergence between official and real-world  $CO_2$  emission values has important implications for fuel efficiency policies. If on-road  $CO_2$  emissions routinely exceed regulated metrics, policies that are solely based on official  $CO_2$  values will fail to achieve the full real-world benefits in terms of fuel savings, climate change mitigation, and reducing reliance on oil products. This problem is particularly acute when the gap is allowed to grow, since benefits of  $CO_2$  or fuel economy standards will be diluted over time.

In order to reflect the real-world performance of the new vehicle fleet in each region, we applied annual gap estimates to the official laboratory measurements in the past years. Because the real-world gap estimates are based on  $CO_2$  emissions values over the official test cycle used by each region, we apply the divergence estimates to certified  $CO_2$  emission values and then normalize the resulting emission values to NEDC equivalents to ensure comparable results (see Kühlwein et al., 2014).

Figure 29 displays the official and real-world average  $CO_2$  emission values for the four markets under study. The EU and China appear to be making little progress in reducing on-road  $CO_2$  emission values after 2008, when the gap started to rapidly grow in both markets. According to the data at hand, the United States has the highest on-road  $CO_2$  emission values, but is improving at a higher rate (1% annual reduction since 2008) than the EU (<1% annual reduction) and China (0% annual reduction). Japan stands out

with the lowest official and real-world  $CO_2$  emission values due to a light, efficient fleet and a comparatively low growth in the gap, although the e-nenpi.com data indicates that the gap has been growing since 2008. On the whole, the figure illustrates that the growing gap is a substantial obstacle to reducing  $CO_2$  emission values in the real world, particularly in the EU and in China, and must therefore be factored in when designing  $CO_2$  or fuel economy standards, a subject that is explored in the next section.

![](_page_53_Figure_2.jpeg)

**Figure 29.** Official and real-world CO<sub>2</sub> emission values for new passenger cars in the EU, the United States, China, and Japan.

Figure 30 shows the development of official and real-world  $CO_2$  values relative to 2008 levels. Japan achieved the highest reduction in official (-27% change) and real-world (-14% change) values. The EU achieved the second highest reduction in official  $CO_2$  values (-20% change), but was outperformed by the United States in terms of real-world performance (-8% change for the United States, -7% change for the EU). Lastly, China achieved a comparatively modest reduction in official values (-8% change) and saw a growth in real-world values (+2% change).

![](_page_54_Figure_1.jpeg)

**Figure 30.** Change in official and real-world  $CO_2$  emission values of new vehicles since 2008 in the EU, the United States, China, and Japan.

# 5.4 GOOD PRACTICES FOR VEHICLE TESTING AND POLICY FRAMEWORKS

Laboratory tests of vehicles cannot produce a single fuel consumption estimate that is accurate for all drivers, but the analysis indicates that some procedures and policy frameworks are better at ensuring representative and consistent results.

Policy enforcement and continuous monitoring of test results seem to be key to ensure that car manufacturers follow the intent and spirit of regulations rather than exploiting loopholes and technicalities. For instance, as the market with the most rigorous policy enforcement, the United States has seen the smallest increase in the gap over time compared to the other three regions and the annual rate of increase has actually slowed a bit since the government adopted new standards. The United States is the only market to test vehicles throughout their useful lives, conduct confirmatory road-load testing, and independently retest a portion of all vehicle models. In contrast, the EU has made substantial improvement in vehicle efficiency on paper, but there is evidence to suggest that a large portion of these improvements resulted from manufacturers gaming vehicle tests and exploiting systemic faults in the type-approval framework (Kadijk et al., 2012; Stewart et al., 2015). Similarly, in the absence of rigorous market oversight, Japan has seen an increase in the gap after fuel economy standards and incentives for efficient vehicles were introduced around 2008. However, these observations are based on limited data, and Section 5.2.1 indicates that the decrease over time in official fuel consumption values may have contributed to the growing gap in Japan. More research is needed to identify the reasons for that growing gap.

Another key feature of the different policies is to what extent they account for the divergence. The impact assessment (European Commission, 2012a, 2012b) accompanying the 2020  $CO_2$  standard assumes a 19.5% gap between real-world and official  $CO_2$  emission values. This gap estimate was based on a 2006 study. Since then, empirical evidence points to significant growth in the gap, reaching 42% for new passenger cars in 2015. This development indicates that there is need to limit the growth

of the gap or to periodically reassess the adjustment factor. In contrast, the adjustment factor used by the United States has been increasing over time as the gap increased. Greene (2015) observes that "because the rulemaking calculated the benefits of the standards assuming a 20% [fuel economy] shortfall, the implication for the projected benefits is very small provided that the gap does not continue to grow" (p. 22). Factoring in the gap during policy design is a low-cost measure to ensure that policies do not overestimate the benefit of on-paper fuel consumption improvements, and could thus help to set economically efficient CO<sub>2</sub> or fuel economy targets.

Measuring the on-road performance of vehicles enables policy actions to manage or close the divergence between official and real-world CO<sub>2</sub> values. The best practice example is the MyMPG service on FuelEconomy.gov, a tool that collects real-world fuel consumption data from consumers. The United States is the only region that has a governmental website for this purpose; the data from Europe, China, and Japan relies on private services. This kind of service allows regulatory agencies to monitor the divergence without relying on third-party data or research. In addition, FuelEconomy.gov presents MyMPG on-road measurements and U.S. EPA label values to the public and therefore helps consumers make informed vehicle purchase decisions.

The design of test cycles and test procedures also deserves attention. The basically non-existent gap for U.S. EPA label values in 2014 indicates that it is possible to design test procedures that, on average, deliver realistic results. In contrast to all other test procedures studied in the report, the five-cycle method measures fuel consumption during aggressive driving, includes the use of air conditioning systems, and tests vehicles at low ambient temperatures. As a result, these values provide reliable estimates of on-road fuel economy for consumers. Nevertheless, while test cycles are frequently cited as a reason for the existence of a gap (e.g., Ding et al., 2015; Kudoh, 2012; Zhang et al., 2014), the foregoing good practices indicate-perhaps somewhat counter-intuitively-that the level of the real-world divergence of regulated  $CO_2$  values is not essential, as long as (1) the gap and any increase in the gap is taken into account when setting  $CO_2$  or fuel economy targets, and (2) consumers have access to reliable fuel consumption values.

## 5.5 LIMITATIONS AND FUTURE RESEARCH

This study covers four major global vehicle markets, each with different vehicle characteristics, unique policies, and varying data availability. The analysis used the best possible data to estimate the divergence between official and on-road CO<sub>2</sub> emission values. Results were discussed with experts from each region to ensure that policy descriptions and real-world data represent each market as accurately as possible. Nonetheless, some limitations remain and are acknowledged below.

Different dating conventions were used in the analyzed data. European samples include vehicle build year (the year a vehicle was constructed), model year (the year a model generation was introduced), fleet year (the year of measurements for an entire fleet), and test year (the year a vehicle was tested). The other data sources, covering China, the United States, and Japan, all use model year to date vehicles. The somewhat inconsistent dating conventions impede like-for-like comparisons of samples, but comparisons between European data sources indicate that these differences do not significantly impact the results. For instance, comparing Spritmonitor.de data, a sample covering German cars by build year, and honestjohn.co.uk, a British data source using model year, reveals that gap estimates from model years tend to be more erratic, but that the general levels of the gap are similar. More importantly, the upward trend in divergence estimates is not affected by the dating convention.

Vehicle definitions vary in the four regions. The analysis generally focuses on passenger cars, and other types of vehicles (light-commercial vehicles in the EU, light trucks in the United States, commercial cargo and passenger vehicles in China, and trucks in Japan) were excluded from comparison charts such as Figure 25 and Figure 27. We attempted to select comparable subsets of the light-duty vehicles by selecting vehicles with similar technical characteristics (passenger cars), but this decision inherently compromises the comparability in terms of how vehicles are used. For instance, the majority of light trucks registered in the United States are used for personal transportation, but U.S. vehicle definitions are based on vehicle characteristics rather than the intended use of a vehicle. In contrast, such vehicles could be registered as passenger cars or light commercial vehicles in the EU, depending on the intended use of the vehicle. While these differences would be significant in comparing technical parameters of the national samples, these differences are less important for the comparison of divergence estimates since the samples indicate that all light-duty vehicles generally follow the same trend (see Figure 17 for example). A study on light commercial vehicles in the EU also suggests that light commercial vehicles follow similar trends as passenger cars (Zacharof, Tietge, Franco, & Mock, 2016). The differing vehicle definitions therefore hamper like-for-like comparisons, but these differences are unlikely to substantially change the divergence estimates and the results of the study.

Limitations in the data collection method should also be acknowledged. The majority of the data-with some exceptions in the EU datasets-were based on fuel consumption values entered by users of web services. This sample design may introduce self-selection bias. Nonetheless, comparisons in each region's analysis section indicate that all realworld samples, with the exception of the Japanese e-nenpi.com data, match vehicle registrations data in terms of vehicle characteristics. Moreover, two samples, Spritmonitor. de and FuelEconomy.gov, provide self-reported information on driver behavior. This data indicates that users are not biased toward driving economically or aggressively, but that most users tend to gravitate to balanced driving styles (Greene et al., 2015; Tietge et al., 2015). Nevertheless, because this data is self-reported and not available for all samples, the potential for skewed samples due to self-selection bias remains.

Taken together, the foregoing limitations imply that the data presented in this study are more suitable for gauging long-term trends in the divergence rather than comparing the different markets in terms of the precise level of the gap in a specific year. It is also the upward trend in the gap that deserves attention in future research. In the EU, a number of studies more or less qualitatively discuss the reasons for the increasing divergence, but quantitative models (e.g., Mellios et al., 2011; Ntziachristos et al., 2014) fail to account for the increase in the gap over time (see Tietge et al., 2017). The findings also echo the call of Greene et al. (2015) for further research on reasons for the increase in the gap in the United States, as well as in other markets. There is also a need to collect more real-world  $CO_2$  emissions data, particularly in markets with comparatively small samples sizes in this study. This is most notable for Japan, where the limited sample size, limitations to the representativeness of the data, and the fact that different test cycles were used during the studied time frame, stand in the way of drawing detailed conclusions from the data at hand. More data could address issues of representativeness of the samples and could help validate the findings presented here.

Lastly, this work has implications for ongoing regulatory discussions in U.S. and EU CO<sub>2</sub> regulations on off-cycle credits. U.S. regulations, within the off-cycle credit provisions, allow approved technologies and additional automaker petitions for new technologies to receive additional credits when they provide evidence of real-world benefits. The U.S. EPA continues to receive and approve petitions (e.g., U.S. EPA, 2015a). The EU regulations, under the eco-innovation provisions, similarly allow technology credits for technologies that deliver greater benefits in the real world than on paper. The results

from this study suggest that automakers are deploying technologies, for instance stop/ start systems, that deliver less real-world benefits than measured during laboratory tests. This calls into question the very logic that automakers can apply for additional credits for the technologies that deliver additional real-world benefits, when a sizeable portion of technologies do not deliver the full extent of benefits during real-world use. We therefore call for more rigorous analyses of how different technologies affect the real-world gap. Moreover, as long as the real-world gap is increasing in some markets, the issuance of off-cycle credits should be reconsidered.

# 6. CONCLUSIONS AND RECOMMENDATIONS

This study analyzes real-world fuel consumption data for more than 1.5 million vehicles from four major vehicle markets, namely the EU, the United States, China, and Japan. The results indicate that the growing divergence between official and real-world  $CO_2$  emission values is a global problem, but that the causes and management of the gap differ from region to region.

This study does not attempt to provide a comprehensive explanation for the growth in the divergence in different regions, but the results indicate that independent confirmatory testing and policy enforcement affect the gap. Lax enforcement in the EU was accompanied by the largest growth in the divergence among the studied regions. In contrast, the United States has the most rigorous compliance and enforcement programs and experienced the smallest growth in the divergence.

Based on the comparison of the divergence and fuel efficiency policies in the different regions, the following aspects should be considered when designing  $CO_2$  or fuel economy standards:

- Independent retesting: Independent retesting of laboratory measurements was identified as a best practice. All markets have some form of compliance program in place, but the United States has the most extensive program, covering the full lifetime of vehicles by verifying coastdown measurements, testing production line vehicles, and conducting in-use surveillance tests.
- » Policy enforcement: The comparatively low growth in the U.S. gap indicates that stringent policy enforcement, such as levying penalties on manufacturers that misstate fuel economy values, acts as a deterrent to gaming. In contrast, the EU has seen the largest growth in the gap from 2001 to 2014 and lacks a central authority to issue vehicle recalls and to impose financial penalties.
- Real-world standards: CO<sub>2</sub> and fuel economy standards should be based on test values that, on average, correspond to real-world measurements. Policies that fail to account for the divergence will overestimate fuel savings and climate change mitigation benefits. Using an adjustment factor to approximate on-road values, as is done in the United States, is an approach that does not require extensive overhauls of vehicle testing procedures in order to account for real-world CO<sub>2</sub> emissions. Another approach for measuring on-road emissions is using PEMS equipment. On-road tests using PEMS are currently only being conducted for nitrogen oxides (NO<sub>x</sub>) and particulate number emissions as part of the RDE procedure, but these kinds of measurements could be used to monitor on-road CO<sub>2</sub> emissions and to test the on-road conformity of vehicles in use. More realistic test cycles and more rigorous testing procedures (e.g., including auxiliary equipment, using stock tires, and using standard engine and transmission calibrations) for laboratory testing could also furnish more realistic CO<sub>2</sub> values.
- Real-world measurements: Measuring real-world fuel consumption is a key recommendation since this data is needed to evaluate the efficacy of CO<sub>2</sub> and fuel economy standards. Bulk on-road fuel consumption data can be measured using web services. As the only government-run example of such services, the MyMPG tool on FuelEconomy.gov stands out as a best practice example. This data can be used to estimate fleet-wide real-world CO<sub>2</sub> emission values and gauge policy impacts. Using data loggers connected to the on-board diagnostics port of vehicles is another option for real-world fuel consumption data collection (see Posada & German, 2013).
- Consumer information: Consumers need access to realistic fuel consumption values to make informed vehicle purchasing decisions. U.S. EPA window label values demonstrate that it is possible to produce fuel consumption values that, on average,

are representative of real-world performance. The FuelEconomy.gov website stands out as a best practice example of consumer information since it combines realworld measurements, realistic fuel consumption values, and information on efficient driving in one portal.

The growing divergence between official and on-road  $CO_2$  emission values is troubling because it represents a decoupling of regulated metrics and real-world impacts. Nevertheless, the recommendations presented here illustrate that solutions are available to close or at least manage the gap.

# REFERENCES

- DieselNet. (2013). United States: GHG emissions & fuel economy. Retrieved from https://www.dieselnet.com/standards/us/fe\_ghg.php
- DieselNet. (2016). Emission Test Cycles [DieselNet]. Retrieved March 10, 2016, from <a href="http://dieselnet.com/standards/cycles/">http://dieselnet.com/standards/cycles/</a>
- Ding, D., Ben Dror, M., Kang, L., & An, F. (2015). *Real-world and certified fuel consumption gap anlaysis*. Beijing: The Innovation Center for Energy and Transportation. Retrieved from http://www.efchina.org/Attachments/Report/report-ctp-20150810/real-world-and-certified-fuel-consumption-gap-analysis
- EEA. (2015). *EU fuel quality monitoring 2014: Summary report* (Publication). European Environment Agency. Retrieved from <a href="http://www.eea.europa.eu/publications/eu-fuel-quality-monitoring-2014">http://www.eea.europa.eu/publications/eu-fuel-quality-monitoring-2014</a>
- European Commission. (2012a). Impact assessment accompanying the documents proposal for a regulation of the European Parliament and of the Council amending Regulation (EC) No 443/2009 to define the modalities for reaching the 2020 target to reduce CO<sub>2</sub> emissions from new passenger cars - part I. Retrieved from http://eur-lex. europa.eu/resource.html?uri=cellar:70f46993-3c49-4b61-ba2f-77319c424cbd.0001.02/ DOC\_2&format=PDF
- European Commission. (2012b). Impact assessment accompanying the documents proposal for a regulation of the European Parliament and of the Council amending Regulation (EC) No 443/2009 to define the modalities for reaching the 2020 target to reduce CO<sub>2</sub> emissions from new passenger cars part II. Retrieved from <a href="http://eur-lex.europa.eu/resource.html?uri=cellar:70f46993-3c49-4b61-ba2f-77319c424cbd.0001.02/DOC\_1&format=PDF">http://eur-lex.europa.eu/resource.html?uri=cellar:70f46993-3c49-4b61-ba2f-77319c424cbd.0001.02/DOC\_1&format=PDF</a>
- European Commission. (2014). Regulation (EU) No 333/2014 of the European Parliament and of the Council of 11 March 2014 amending Regulation (EC) No 443/2009 to define the modalities for reaching the 2020 target to reduce CO2 emissions from new passenger cars. Retrieved from http://eur-lex.europa.eu/legal-content/EN/TXT/PDF/?u ri=CELEX:32014R0333&from=EN
- European Commission. (2016a). A European strategy for low-emission mobility. Brussels: European Commission. Retrieved from http://ec.europa.eu/transport/themes/ strategies/news/doc/2016-07-20-decarbonisation/com(2016)501\_en.pdf
- European Commission. (2016b). Proposal for a regulation of the European Parliament and of the Council on the approval and market surveillance of motor vehicles and their trailers, and of systems, components and separate technical units intended for such vehicles. Brussels: European Commission. Retrieved from http://eur-lex.europa.eu/ legal-content/EN/TXT/?qid=1453996101816&uri=COM:2016:31:FIN
- European Commission. (2016c). Reducing CO2 emissions from passenger cars. Retrieved September 14, 2016, from <u>http://ec.europa.eu/clima/policies/transport/vehicles/cars/</u> <u>index\_en.htm</u>
- European Parliament. (2007). Directive 2007/46/EC of the European Parliament and of the Council of 5 September 2007 establishing a framework for the approval of motor vehicles and their trailers, and of systems, components and separate technical units intended for such vehicles. Retrieved from http://eur-lex.europa.eu/legal-content/EN/ TXT/?uri=celex:32007L0046
- Franco, V. (2016). *Proposed new type-approval system for motor vehicles in the European Union*. The International Council on Clean Transportation. Retrieved from <a href="http://www.theicct.org/proposed-new-type-approval-framework-eu-policy-update">http://www.theicct.org/proposed-new-type-approval-framework-eu-policy-update</a>

- Franco, V., & Mock, P. (2015). *The European Real-Driving Emissions regulation*. The International Council on Clean Transportation. Retrieved from <a href="http://www.theicct.org/sites/default/files/ICCT\_facts\_EU-emissions-testing\_jun2016.pdf">http://www.theicct.org/sites/default/files/ICCT\_facts\_EU-emissions-testing\_jun2016.pdf</a>
- German, J. (2015). *Hybrid vehicles Technology development and cost reduction*. The International Council on Clean Transportation. Retrieved from <a href="http://www.theicct.org/sites/default/files/publications/ICCT\_TechBriefNo1\_Hybrids\_July2015.pdf">http://www.theicct.org/sites/default/files/publications/ICCT\_TechBriefNo1\_Hybrids\_July2015.pdf</a>
- Greene, D. L., Goeltz, R., Hopson, J. L., & Tworek, E. (2007). Analysis of In-Use Fuel Economy Shortfall Based on Voluntarily Reported MPG Estimates. In *Transportation Research Record* (Vol. 1983, pp. 99–105). Retrieved from <u>http://www.osti.gov/scitech/</u> biblio/931042
- Greene, D. L., Khattak, A., Liu, J., Hopson, J. L., Wang, X., & Goeltz, R. (2015). How do motorists' own fuel economy estimates compare with official government ratings? A statistical analysis (Baker Reports No. 4:15). Knoxville: The Howard H. Baker Jr. Center for Public Policy. Retrieved from http://bakercenter.utk.edu/wp-content/uploads/ sites/4/2015/09/BR\_Greene\_4-15.pdf
- He, H., & Tu, J. (2012). *The new passenger car fleet in China, 2010*. The International Council on Clean Transportation. Retrieved from <a href="http://www.theicct.org/new-passenger-car-fleet-china-2010">http://www.theicct.org/new-passenger-car-fleet-china-2010</a>
- He, H., & Yang, Z. (2014a). *China Phase 3 new passenger car fuel consumption: proposed implementation rule*. The International Council on Clean Transportation. Retrieved from http://www.theicct.org/china-phase-3-new-passenger-car-fuel-consumption-proposed-implementation-rule
- He, H., & Yang, Z. (2014b). *China Phase 4 passenger car fuel consumption standard proposal*. The International Council on Clean Transportation. Retrieved from <a href="http://www.theicct.org/china-phase-4-passenger-car-fuel-consumption-standard-proposal">http://www.theicct.org/china-phase-4-passenger-car-fuel-consumption-standard-proposal</a>
- Huo, H., Yao, Z., He, K., & Yu, X. (2011). Fuel consumption rates of passenger cars in China: Labels versus real-world. *Energy Policy*, *39*(11), 7130–7135. <u>https://doi.org/10.1016/j.enpol.2011.08.031</u>
- ICCT. (2016a). Global passenger vehicle standards. Retrieved August 16, 2016, from http://www.theicct.org/info-tools/global-passenger-vehicle-standards
- ICCT. (2016b). Global PV standards chart library. Retrieved March 10, 2016, from http://theicct.org/global-pv-standards-chart-library
- ICCT. (2016c). Vehicle emissions testing the European Union. The International Council on Clean Transportation. Retrieved from <a href="http://www.theicct.org/sites/default/files/liccT%20handout%20Final\_3.pdf">http://www.theicct.org/sites/default/files/ ICCT%20handout%20Final\_3.pdf</a>
- JAMA. (2015). *The motor industry of Japan 2015* (p. 67). Tokyo, Japan: Japan Automobile Manufacturers Association, Inc. Retrieved from <u>http://www.jama-english.jp/</u> <u>publications/MIJ2015.pdf</u>
- Kadijk, G., Verbeek, M., Smokers, R., Spreen, J., Patuleia, A., van Ras, M., ... Buttigieg, D. (2012). Supporting Analysis regarding Test Procedure Flexibilities and Technology Deployment for Review of the Light Duty Vehicle CO2 Regulations (No. 033.22993).
   TNO. Retrieved from <a href="http://ec.europa.eu/clima/policies/transport/vehicles/cars/docs/report\_2012\_en.pdf">http://ec.europa.eu/clima/policies/transport/vehicles/cars/docs/report\_2012\_en.pdf</a>
- Kudoh, Y. (2012). Understanding Fuel Consumption/Economy of Passenger Vehicles in the Real World. In *Internal Combustion Engines* (Kazimierz Lejda and Pawel Wos). InTech. Retrieved from <u>http://www.intechopen.com/books/internal-combustion-engines/</u> understanding-fuel-consumption-economy-of-passenger-vehicles-in-the-real-world

- Kühlwein, J. (2016). Official vs. real-world road-load parameters in EU vehicle efficiency testing. The International Council on Clean Transportation. Retrieved from http://www.theicct.org/effect-roadload-coeffs-co2-emissions-eu
- Kühlwein, J., German, J., & Bandivadekar, A. (2014). Development of Test Cycle Conversion Factors among Worldwide Light-Duty Vehicle CO2 Emission Standards.
   International Council on Clean Transportation. Retrieved from <a href="http://www.theicct.org/sites/default/files/publications/ICCT\_LDV-test-cycle-conversion-factors\_sept2014.pdf">http://www.theicct.org/sites/default/files/publications/ICCT\_LDV-test-cycle-conversion-factors\_sept2014.pdf</a>
- Ligterink, N. E., & Eijk, A. (2014). Update analysis of real-world fuel consumption of business passenger cars based on Travelcard Nederland fuelpass data. Delft: TNO. Retrieved from <u>https://www.tno.nl/content.cfm?context=thema&content=inno\_publicat</u> ie&laag1=896&laag2=913&item\_id=990
- Ligterink, N. E., & Smokers, R. T. M. (2015). Monitoring van plug-in hybride voertuigen (PHEVs) april 2012 t/m maart 2015. Retrieved January 11, 2016, from http://repository. tudelft.nl/view/tno/uuid%3A72869589-afaf-41a9-9e1a-2dd0d0f05061/
- Ligterink, N., Smokers, R. T., Spreen, J., Mock, P., & Tietge, U. (2016). *Supporting analysis* on real-world light-duty vehicle CO2 emissions (No. TNO 2016 R10419v3). Delft, the Netherlands: TNO. Retrieved from <u>http://ec.europa.eu/clima/policies/transport/</u> vehicles/docs/analysis\_ldv\_co2\_emissions\_en.pdf
- Mellios, G., Hausberger, S., Keller, M., Samaras, C., & Ntziachristos, L. (2011). Parametrization of fuel consumption and CO<sub>2</sub> emissions of passenger cars and light commercial vehicles for modelling purposes.
- Mock (ed.), P. (2015). *European vehicle market statistics Pocketbook 2015/16*. The International Council on Clean Transportation. Retrieved from <a href="http://eupocketbook.theicct.org/">http://eupocketbook.theicct.org/</a>
- Mock, P. (2014). EU CO2 standard for passenger cars and light-commercial vehicles. International Council on Clean Transportation. Retrieved from http://www.theicct.org/ eu-co2-standards-passenger-cars-and-lcvs
- Mock, P., & German, J. (2015). *The future of vehicle emissions testing and compliance* (White Paper). International Council on Clean Transportation. Retrieved from <a href="http://theicct.org/future-of-vehicle-testing">http://theicct.org/future-of-vehicle-testing</a>
- Mock, P., German, J., Bandivadekar, A., & Riemersma, I. (2012). Discrepancies between type-approval and real-world fuel consumption and CO2 values in 2001-2011 European passenger cars. Washington: The International Council on Clean Transportation. Retrieved from http://theicct.org/fuel-consumption-discrepancies
- Mock, P., Tietge, U., Franco, V., German, J., Bandivadekar, A., Ligterink, N., ... Riemersma,
   I. (2014). From laboratory to road: A 2014 update of official and "real-world" fuel consumption and CO2 values for passenger cars in Europe. Retrieved from http://www.theicct.org/laboratory-road-2014-update
- Næss-Schmidt, H. S., & Winiarczyk, M. (2010). *Company car taxation: subsidies, welfare and environment*. Luxembourg: EUR-OP.
- NHTSA. (2014). Summary of fuel economy performance. Washington DC: U.S. Department of Transportation. Retrieved from file:///Users/uwetietge/Downloads/ Performance-summary-report-12152014-v2.pdf
- Ntziachristos, L., Mellios, G., Tsokolis, D., Keller, M., Hausberger, S., Ligterink, N., & Dilara, P. (2014). In-use vs. type-approval fuel consumption of current passenger cars in Europe. *Energy Policy*, 67, 403–411. https://doi.org/10.1016/j.enpol.2013.12.013
- Posada, F., & German, J. (2013). *Measuring in-use fuel economy: Summary of pilot studies*. The International Council on Clean Transportation. Retrieved from http://www.theicct.org/measuring-in-use-fuel-economy-summary-pilot-studies

- Qin, L., Ben Dror, M., Kang, L., Sun, H., & An, F. (2016). *Real-world and certified fuel consumption gap anlaysis 2016*. Beijing: The Innovation Center for Energy and Transportation. Retrieved from <u>http://www.icet.org.cn/admin/upload/2016092336232621.pdf</u>
- Regulation (EC) No 443/2009. (2009). Regulation (EC) No 443/2009 of the European Parliament and of the Council of 23 April 2009 setting emission performance standards for new passenger cars as part of the Community's integrated approach to reduce CO2 emissions from light-duty vehicles. Retrieved from http://eur-lex.europa. eu/legal-content/EN/TXT/HTML/?uri=CELEX:02009R0443-20130508&from=EN
- Rutherford, D. (2014, April 4). Hybrids break through in the Japan auto market. Retrieved from http://www.theicct.org/blogs/staff/hybrids-break-through-japan-auto-market
- Rutherford, D. (2015, July 28). Hybrids break into the Japanese market (July 2015 update). Retrieved from <a href="http://www.theicct.org/blogs/staff/hybrids-break-japanese-market-july-2015-update">http://www.theicct.org/blogs/staff/hybrids-break-japanese-market-july-2015-update</a>
- Schipper, L. (2008). Automobile fuel; economy and CO2 emissions in industrialized countries: Troubling trends through 2005/6. Washington, D.C.: EMBARQ. Retrieved from http://pdf.wri.org/automobile-fuel-economy-co2-industrialized-countries.pdf
- Schipper, L., Figueroa, M. J., Price, L., & Espey, M. (1993). Mind the gap The vicious circle of measuring automobile fuel use. *Energy Policy*, 21(12), 1173–1190. <u>https://doi.org/10.1016/0301-4215(93)90268-K</u>
- Schipper, L., & Tax, W. (1994). New car test and actual fuel economy: yet another gap? *Transport Policy*, 1(4), 257–265. https://doi.org/10.1016/0967-070X(94)90006-X
- Smokers, R., Vermeulen, R., van Mieghem, R., Gense, R., Skinner, I., Fergusson, M., ... Samaras, Z. (2006). Review and analysis of the reduction potential and costs of technological and other measures to reduce CO2-emissions from passenger cars. *TNO Report*, 6. Retrieved from http://edz.bib.uni-mannheim.de/daten/edz-h/gdb/06/ report\_co2\_reduction.pdf
- Soble, J. (2016, May 11). Mitsubishi scandal expands to more models. Retrieved from http://www.nytimes.com/2016/05/12/business/international/mitsubishi-cheatingscandal-expands-to-more-models.html?\_r=0
- Stewart, A., Hope-Morley, A., Mock, P., & Tietge, U. (2015). Impact of real-world driving on emissions from UK cars and vans. Cambridge: Committee on Climate Change. Retrieved from https://www.theccc.org.uk/publication/impact-of-real-world-driving-emissions/
- Tietge, U., Díaz, S., Mock, P., German, J., Bandivadekar, A., & Ligterink, N. (2016). From laboratory to road - A 2016 update of official and "real-world" fuel consumption and CO2 values for passenger cars in Europe. The International Council on Clean Transportation. Retrieved from http://www.theicct.org/laboratory-road-2016-update
- Tietge, U., Mock, P., Franco, V., & Zacharof, N. (2017). From laboratory to road: Modeling the divergence between official and real-world fuel consumption and CO2 emission values in the German passenger car market for the years 2001–2014. *Energy Policy*, 103, 212–222. https://doi.org/10.1016/j.enpol.2017.01.021
- Tietge, U., Zacharof, N., Mock, P., Franco, V., German, J., Bandivadekar, A., ... Lambrecht, U. (2015). From laboratory to road - A 2015 update of official and "real-world" fuel consumption and CO<sub>2</sub> values for passenger cars in Europe. The International Council on Clean Transportation. Retrieved from http://theicct.org/laboratory-road-2015-update
- Tu, J., Zou, W., & He, H. (2014). New light commercial vehicles in China, 2010. The International Council on Clean Transportation. Retrieved from <u>http://www.theicct.org/</u> <u>china-lcvs-2010-assessment</u>

- UNECE. (2013). Addendum 100: Regulation No. 101. United Nations. Retrieved from https:// www.unece.org/fileadmin/DAM/trans/main/wp29/wp29regs/updates/R101r3e.pdf
- U.S. Environmental Protection Agency. (2006). *Fuel economy labeling of motor vehicle: Revisions to improve calculation of fuel economy estimates*. U.S. Environmental Protection Agency. Retrieved from <a href="http://www3.epa.gov/carlabel/documents/420r06017.pdf">http://www3.epa.gov/carlabel/documents/420r06017.pdf</a>
- U.S. EPA. (2006). *Fuel economy labeling of motor vehicle: Revisions to improve calculation of fuel economy estimates*. U.S. Environmental Protection Agency. Retrieved from http://www3.epa.gov/carlabel/documents/420r06017.pdf
- U.S. EPA. (2008). 2007 progress report vehicle and engine compliance activities. Retrieved from http://www.epa.gov/otaq/about/420r08011.pdf
- U.S. EPA. (2010a). 40 CFR Parts 85, 86, and 600; 49 CFR Parts 531, 533, 536, et al. Retrieved from https://www.gpo.gov/fdsys/pkg/FR-2010-05-07/pdf/2010-8159.pdf
- U.S. EPA. (2010b). *EPA and NHTSA finalize historic national program to reduce greenhouse gases and improve fuel economy for cars and trucks*. Retrieved from https://www3.epa.gov/otaq/climate/regulations/420f10014.pdf
- U.S. EPA. (2010c, April). Final rulemaking to establish light-duty vehicle greenhouse gas emission standards and corporate average fuel economy standards Regulatory impact assessment. Retrieved from <a href="https://www3.epa.gov/otag/climate/">https://www3.epa.gov/otag/climate/</a> regulations/420r10009.pdf
- U.S. EPA. (2014a, June 12). Fuel Economy Label: Updates [Overviews & Factsheets]. Retrieved September 12, 2016, from https://www3.epa.gov/fueleconomy/updates.htm
- U.S. EPA. (2014b, November 3). Hyundai and Kia Clean Air Act Settlement [Policies and Guidance]. Retrieved September 12, 2016, from <a href="https://www.epa.gov/enforcement/hyundai-and-kia-clean-air-act-settlement">https://www.epa.gov/enforcement/hyundai-and-kia-clean-air-act-settlement</a>
- U.S. EPA. (2015a). *EPA decision document: Off-cycle credits for Fiat Chrysler Automobiles, Ford Motor Company, and General Motors Corporation*. U.S. Environmental Protection Agency. Retrieved from https://nepis.epa.gov/Exe/ZyNET. exe/P100N19E.TXT?ZyActionD=ZyDocument&Client=EPA&Index=2011+Thru+2015&Doc s=&Query=&Time=&EndTime=&SearchMethod=1&TocRestrict=n&Toc=&TocEntry=&QFiel d=&QFieldYear=&QFieldMonth=&QFieldDay=&IntQFieldOp=0&ExtQFieldOp=0&XmlQu ery=&File=D%3A%5Czyfiles%5CIndex%20Data%5C11thru15%5CTxt%5C00000016%5CP 100N19E.txt&User=ANONYMOUS&Password=anonymous&SortMethod=h%7C-&Maxim umDocuments=1&FuzzyDegree=0&ImageQuality=r75g8/r75g8/x150y150g16/i425&Dis play=hpfr&DefSeekPage=x&SearchBack=ZyActionL&Back=ZyActionS&BackDesc=Resu Its%20page&MaximumPages=1&ZyEntry=1&SeekPage=x&ZyPURL
- U.S. EPA. (2015b). *Light-duty automotive technology, carbon dioxide emissions, and fuel economy trends: 1975 through 2015.* U.S. Environmental Protection Agency. Retrieved from https://www3.epa.gov/fueleconomy/fetrends/1975-2015/420r15016.pdf
- U.S. EPA. (2015c, June 22). Derived 5-cycle coefficients for 2017 and later model years. Retrieved from https://iaspub.epa.gov/otaqpub/display\_file.jsp?docid=35113&flag=1
- U.S. EPA, CARB, & NHTSA. (2016). Draft technical assessment report: Midterm evaluation of light-duty vehicle greenhouse gas emission standards and corporate average fuel economy standards for model years 2022-2025. Retrieved from <a href="https://www3.epa.gov/otag/climate/documents/mte/420d16900.pdf">https://www3.epa.gov/otag/climate/documents/mte/420d16900.pdf</a>
- Yang, Z., & Muncrief, R. (2017). *Market surveillance of vehicle emissions: Best-practice examples with respect to the European Commission's proposed type-approval framework regulation*. The International Council on Clean Transportation. Retrieved from http://theicct.org/market-surveillance-of-vehicle-emissions

- Zacharof, N., Tietge, U., Franco, V., & Mock, P. (2016). Type approval and real-world CO<sub>2</sub> and NO<sub>x</sub> emissions from EU light commercial vehicles. *Energy Policy*, *97*, 540–548. https://doi.org/10.1016/j.enpol.2016.08.002
- Zhang, S., Wu, Y., Liu, H., Huang, R., Un, P., Zhou, Y., ... Hao, J. (2014). Real-world fuel consumption and CO2 (carbon dioxide) emissions by driving conditions for lightduty passenger vehicles in China. *Energy*, 69, 247–257. <u>https://doi.org/10.1016/j.energy.2014.02.103</u>

# APPENDIX I: FUEL ECONOMY CHARTS

![](_page_66_Figure_2.jpeg)

**Figure 31.** Divergence between MyMPG and official fuel economy values for light-duty vehicles and cars. Official fuel economy values include CAFE figures and EPA label values.

![](_page_66_Figure_4.jpeg)

**Figure 32.** Divergence of MyMPG from official fuel economy values by powertrain. Official fuel economy values include CAFE figures and EPA label values.

![](_page_67_Figure_1.jpeg)

Figure 33. Divergence between official and e-nenpi.com fuel economy values.

# APPENDIX II: CALCULATION METHODOLOGY FOR E-NENPI.COM

Each row of the e-nenpi data set described one year of data for one model version, where a model version is defined as a unique combination of vehicle brand, model name, model year, body style, fuel type, engine displacement, transmission type, number of gears, and other powertrain characteristics. Since different vehicles of a given model version may have entered and left the database over time, and data was delivered aggregated by model version and calendar year, the number of vehicles in the sample was estimated as follows:

 $n_i = \max(n_{i,t=0}, n_{i,t=1}, ..., n_{i,t=t})$ , where

- n, = number of vehicles of model version i
- n<sub>i</sub> = period of data collection
- T = total number of data collection periods

Furthermore, only minimum and maximum type-approval fuel consumption values of each model version were delivered by e-nenpi.com. Type-approval fuel consumption values of model versions were thus estimated as:

TAFC = 
$$\frac{TAFC_{min} + TAFC_{max}}{2}$$
, where

TAFC = type-approval fuel consumption

The average divergence between type-approval and e-nenpi.com fuel consumption values was calculated as the weighted arithmetic mean of the divergence of all model versions:

$$\delta = \frac{\sum_{i=1}^{V} n_i x \delta_i}{\sum_{i=1}^{V} n_i}, \text{ where }$$

V = total number of model version

 $\delta$  = divergence